DOCUMENTOS DE TRABAJO

The impact of financial crises on industrial growth: lessons from the last 40 years

Carlos Madeira

N° 1044 Junio 2025

BANCO CENTRAL DE CHILE







La serie Documentos de Trabajo es una publicación del Banco Central de Chile que divulga los trabajos de investigación económica realizados por profesionales de esta institución o encargados por ella a terceros. El objetivo de la serie es aportar al debate temas relevantes y presentar nuevos enfoques en el análisis de los mismos. La difusión de los Documentos de Trabajo sólo intenta facilitar el intercambio de ideas y dar a conocer investigaciones, con carácter preliminar, para su discusión y comentarios.

La publicación de los Documentos de Trabajo no está sujeta a la aprobación previa de los miembros del Consejo del Banco Central de Chile. Tanto el contenido de los Documentos de Trabajo como también los análisis y conclusiones que de ellos se deriven, son de exclusiva responsabilidad de su o sus autores y no reflejan necesariamente la opinión del Banco Central de Chile o de sus Consejeros.

The Working Papers series of the Central Bank of Chile disseminates economic research conducted by Central Bank staff or third parties under the sponsorship of the Bank. The purpose of the series is to contribute to the discussion of relevant issues and develop new analytical or empirical approaches in their analyses. The only aim of the Working Papers is to disseminate preliminary research for its discussion and comments.

Publication of Working Papers is not subject to previous approval by the members of the Board of the Central Bank. The views and conclusions presented in the papers are exclusively those of the author(s) and do not necessarily reflect the position of the Central Bank of Chile or of the Board members.

Documento de Trabajo Nº 1044

Working Paper N° 1044

The impact of financial crises on industrial growth: lessons from the last 40 years*

Carlos Madeira

Resumen

Este trabajo muestra el impacto de las crisis financieras en diversas industrias manufactureras y en el sector manufacturero en su totalidad. Los resultados muestran tanto un impacto directo de las crisis financieras en el crecimiento de todas las manufacturas, como un efecto adicional a través de un canal de dependencia de financiamiento externo. Las industrias externamente dependientes experimentan un menor crecimiento durante las crisis bancarias y de tipo de cambio, especialmente en mercados emergentes y economías en desarrollo. Las crisis bancarias, de tipo de cambio y de deuda soberana causan una reducción promedio en el crecimiento total de las manufacturas del 2.7%, 6% y 1%, respectivamente, siendo el efecto directo el componente más significativo. Finalmente, demuestro que las políticas macroprudenciales adoptadas después de la Gran Recesión de 2008 atenuaron la caída en el crecimiento causada por las crisis bancarias en la década siguiente.

Abstract

This work shows the impact of financial crises across industries and the total manufacturing sector. I find both a direct impact of financial crises on all manufacturing growth and an additional effect through an external finance dependence channel. Externally dependent industries experience lower growth during banking and currency crises, especially in emerging markets and developing economies. Banking, currency and sovereign debt crises cause an average reduction in total manufacturing growth of 2.7%, 6% and 1%, respectively, with the direct effect being the most significant component. Finally, I show that macroprudential policies adopted after the Great Financial Crisis attenuated the fall in growth caused by banking crises.

^{*}Bank for International Settlements (BIS) and Central Bank of Chile. Comments are welcome at carlosmadeira2009@u.northwestern.edu. For comments, I thank Jon Frost, Christian Upper, Carola Müller, Bernardus Doornik, Alfredo Pistelli, an anonymous referee from the BIS working paper series and seminar participants at the BIS Americas office, LACEA 2024 and at the Financial Market Commission (CMF) in Chile. The views and conclusions in this paper are exclusively those of the author and do not necessarily reflect the position of the BIS or the Central Bank of Chile or its board members. The views expressed are those of the author and do not necessarily reflect the views of the Central Bank of Chile or its board members. All errors are my own.

1 Introduction

Financial crises are periods in which economic activity falls substantially (Schularick and Taylor 2012). The frequency of crises has increased substantially since the end of the Bretton-Woods era (Hoggarth et al. 2002, Bordo and Meissner 2016). This has led many policy makers, particularly in emerging markets and developing economies, to question the advantages of financial liberalization and globalization (Bordo and Meissner 2016, Rodrik 2021). Banking crises are especially important, because these often precede other financial crises such as sovereign debt default, currency crises and sudden stops (Kaminsky and Reinhart 1999, Bordo and Meissner 2016). It is also often the case that currency crises and sudden stops do not have such deep effects unless these coincide with a banking crisis (Joyce and Nabar 2009). At the same time, macroeconomic models have difficulties in accounting for financial frictions and the deepness of crises (Linde et al. 2016, Galati and Moessner 2018) such as the Great Financial Crisis (hence on, GFC). Therefore, further research is needed on financial crisis channels and the policies required to shield the economies from their effects (Akinci and Olmstead-Rumsey 2018, Madeira 2024).

This work studies the impact of financial crises on the real growth of a panel of 23 manufacturing industries across 102 countries for the period between 1980 and 2019. One advantage of using detailed industry data is that each industry is small relative to the national economy, addressing the issues of reverse-causality between growth and national shocks such as banking crises (Madeira 2024).¹ A second advantage of the industry level data is that industries' heterogeneous characteristics and their differential capital needs can inform us about the financial channel in which banking

¹Note that assuming small industries are affected by national shocks, rather than the reverse, is a standard assumption in the macroeconomics literature. For instance, Rajan and Zingales 1998, Braun and Larrain 2005, Kroszner et al. 2007, Dell'Ariccia et al. 2008 and Cowan and Raddatz 2013 assume that national financial development, recessions, banking crises and sudden stops are exogenous relative to the individual manufacturing industries. This assumption is also credible relative to other usual identification assumptions used in the macroeconomics literature. For instance, it is standard to assume that small open economies are affected by international shocks such as oil prices, which are taken to be exogenous due to the small size of each country. There are around 196 countries in the world, with 189 nations being members of the World Bank and IMF. Therefore, the average nation represents just 0.51% of the world's GDP. The median manufacturing industry in this dataset represents just 0.32% of the national GDP (as shown in Table 1). Therefore, these are industries that are too small to influence by themselves large national developments such as a financial crisis.

crises affect non-financial firms' activity (Dell'Ariccia et al. 2008, Igan et al. 2022). However, one important limitation of the study is that manufacturing activity represents just 12.7% of the national GDP for the median country. Therefore, this study presents a reliable estimate of financial crisis impact on a set of industries, but it does not analyze its total effect on economic activity.

This study uses mainly an industry-country dataset from the United Nations Industrial Development Organization (UNIDO), which compiles national accounts information for more than 100 countries using the same set of 23 manufacturing industries. I then combine this data with a list of the banking, currency and sovereign debt crises for the period 1980 to 2019, compiled by Laeven and Valencia 2020 and Nguyen et al. 2022. I estimate banking crisis effects on the industries' growth using simple linear methods, with country-industry and time fixed effects, with further controls such as financial crisis dummies and GDP per capita (as a measure of the countries' development levels). Other consistent estimates of the financial crisis effects on growth are obtained using quantile regression methods for panel data (Machado and Santos-Silva 2019). Furthermore, I analyze the impact of banking crises across different country groups, such as advanced economies (AEs) and emerging markets and developing economies (EMDEs). Finally, I show the results are robust to including country-industry, country-year and industry-year fixed effects.

My methodology estimates both a direct effect of financial crises on all industries and an additional effect through the external finance dependence channel which is heterogeneous across industries.² This is a relevant improvement upon previous studies, such as Dell'Ariccia et al. 2008, which only studied the external finance dependence channel. My results show that the direct effect of financial crises has a larger impact on manufacturing growth. Banking, currency and sovereign debt crises cause a reduction in total manufacturing growth of 2.7%, 6% and 1% for all countries, 2.5%, 4.1% and 3.4% for AEs and 2.1%, 6.1% and 0.3% for EMDEs. Currency crises impact EMDEs more strongly, while sovereign debt crises have stronger effects on AEs. Furthermore, my estimates of the effects of banking crises on manufacturing are much larger than those found in previous studies, such as Dell'Ariccia et al. 2008 which ignored the GFC period.

One interesting issue is the effect of financial regulations taken after the GFC. To obtain the effect of macroprudential policies, I add as controls the macroprudential policy tightening

²The reason is that companies that cannot fund their capital expenses using past earnings must resort to outside funds through banks or capital markets, therefore being more sensitive to financial shocks such as crises.

implemented in each year and the macroprudential policies accumulated since 2010 for each country. I find that macroprudential policies had a positive impact on industrial growth during the banking crises for the period after 2010 across all country samples (all countries, AEs, EMDEs). However, macroprudential policies also reduced the growth of some industries with high external finance dependence, particularly in EMDEs. Furthermore, macroprudential policy easing during banking crises had a positive impact on growth of externally dependent industries for the samples of all countries and AEs. Considering just the external finance dependence channel, banking crises in the post-GFC period would have caused an additional loss of 2.2% in manufacturing growth. However, the direct effect of macroprudential policies during banking crises on all industries has a counterbalancing impact. Therefore, macroprudential policies during banking crises had an overall positive effect on manufacturing growth relative to a scenario in which no policies had been implemented. This beneficial effect of macroprudential policies during banking crises in the post-GFC period happens across all country groups (all countries, AEs, EMDEs). I find that macroprudential policies increased manufacturing growth during banking crises in 4.3%, 2.2% and 8.3% for all countries, AEs and EMDEs. These results confirm the importance of accounting for both the direct effects and the external finance dependence channel of crises.

This study is mostly related to the work of Dell'Ariccia et al. 2008, which shows that, during the period between 1980 and 2000, the sectors more dependent on external finance suffer the most during banking crises. In a closely related paper, Kroszner et al. 2007 find that the relative growth in value added of financially dependent industries is faster in financially developed countries in pre-crisis periods but slower in crisis periods. My work extends the analysis in Dell'Ariccia et al. 2008 by adding the last twenty years, a more extensive list of countries (102 countries instead of 41 countries) and a more detailed analysis of the historical list of banking crises (as obtained from the works of Laeven and Valencia 2020 and Nguyen et al. 2022). This extension of the analysis in the previous studies is especially relevant, because the largest financial crisis in the last 40 years happened in 2008 and was outside the scope of past literature. Furthermore, it was only after the Great Financial Crisis of 2008 that macroeconomic research undertook a large effort to include the modeling of financial frictions (Linde et al. 2016). Moreover, there were several reforms to banking regulation, including the wider introduction of macroprudential policies (Akinci and Olmstead-Rumsey 2018, Galati and Moessner 2018). My study shows that emerging markets

are more sensitive to banking crises, which is consistent with prior literature showing that these countries are more procyclical (Frankel 2010).

Relative to the closest previous study of Dell'Ariccia et al. 2008, this work brings three important contributions: i) it analyzes three times as much data as a combination of more countries and twenty additional years (the additional data substantially increases the estimates of the effects of banking crises in AEs); ii) it estimates the full impact of financial crises, through the sum of its direct effect on manufacturing growth and the external finance dependence channel (it is shown that the direct effect, ignored in previous studies, can be two times as large or more than the external finance dependence channel); iii) I show that macroprudential policies implemented after 2010 had a big effect in attenuating the impact of banking crises.

Finally, this paper complements the literature that shows how external finance dependent industries are more strongly affected by financial development (Rajan and Zingales 1998, Raddatz 2006), recessions (Braun and Larrain 2005), banking crises (Dell'Ariccia et al. 2008), sudden-stops (Cowan and Raddatz 2013) and capital flows (Igan et al. 2020, Igan et al. 2022). The article also complements empirical studies showing that a trade-off between the benefits of financial development on growth (Levine 2005, Levchenko et al. 2009) and the cost of increased volatility or crises (Rancière et al. 2008, Dwyer et al. 2013, Laeven and Valencia 2020). Furthermore, Hoggarth et al. 2002 claim that, contrary to popular belief, output losses associated with banking crises are not more severe in developing countries than in developed countries. My work shows that the effect of financial crises on the externally dependent industries is much stronger in EMDEs than in AE.

This paper is organized as follows. Section 2 describes the data sources and the empirical approach. Section 3 shows the main estimates of the effect of banking crises on industrial growth and the differences across countries' development levels. Section 4 shows the effects of other types of financial crises on industrial growth. Section 5 summarizes all the estimates in terms of the total impact on the national manufacturing sector. Section 6 estimates the effect of macroprudential policies implemented after the GFC on industrial growth during banking crises. Finally, Section 7 concludes with a summary of the findings and its policy implications.

2 Data and empirical approach

2.1 Data

This study uses the UNIDO's Industrial Statistics Database (Indstat2 - revision 3), which contains annual frequency data for the 2-digit ISIC (International Standard Industrial Classification of All Economic Activities) industries of each country for 23 manufacturing industries. Industrial growth is measured by the log increase in the Index of Real Industrial Production (IIP), $g_{i,c,t} = 100 \ln(\frac{IIP_{i,c,t}}{IIP_{i,c,t-1}})$, which accounts for sector-specific prices. Real industrial growth is multiplied by 100 to be similar to the national growth rates in World Bank and IMF data, which are also reported in percentage points. The data are unbalanced, with some missing data.

Table 1 summarizes the list of industries and countries available in the dataset. This study comprises 102 countries, including 35 AEs and 67 EMDEs. The median country produces just 12.7% of its GDP from the manufacturing sector. The data shows that even the largest industry of each country is relatively small. The largest industry of the median country produces just 2.8% of GDP. Less than 10% of the countries have an industry that represents more than 5.1% of GDP. No country has an industry that represents more than 18% of the GDP. Finally, Table 1 shows that the median industry produces just 0.3% of the GDP and has a correlation of just 42% with the national real GDP growth. Therefore, the assumptions required for the identification mechanism in this article are valid: i) all industries are small (the median industry represents just 0.3% of the GDP) and not the main cause of the financial crises;³ ii) the industries are not similar to a "representative firm" agent and have a correlation with real GDP growth that is less than 42%.

³The small value of each individual industry relative to the national economy is crucial to insure that financial crises and other aggregate events can be taken as exogenous. For this dataset, the assumption of the exogeneity of national outcomes is realistic. Assume that the median industry, which represents just 0.3% of national GDP, has a huge crisis and loses one third of its output. Even this large industrial shock represents just 0.1% of GDP, which is unlikely to cause a national financial crisis due to such a small value at the national level. Therefore, it is reasonable to assume that financial crises impact the manufacturing industries, but not the other way around.

Table 1: Industries and countries available in the joint industrial and macroprudential policy dataset Industries (ISIC 2-digit revision 3) with External Finance Dependence (EFD_i) in parentheses: 15 Food and beverages (0.112), 16 Tobacco products (-0.451), 17 Textiles (0.277), 18 Wearing apparel, fur (0.029), 19 Leather, leather products and footwear (-0.113), 20 Wood products (excl. furniture) (0.283), 21 Paper and paper products (0.161), 22 Printing and publishing (0.203), 23 Coke, refined petroleum products, nuclear fuel (0.170), 24 Chemicals and chemical products (0.458), 25 Rubber and plastics products (0.634), 26 Non-metallic mineral products (0.193), 27 Basic metals (0.040), 28 Fabricated metal products (0.213), 29 Machinery and equipment n.e.c. (0.633), 30 Office, accounting and computing machinery (0.948), 31 Electrical machinery and apparatus (0.821), 32 Radio, television and communication equipment (0.975), 33 Scientific instruments, medical, precision and optical instruments (0.961), 34 Motor vehicles, trailers, semi-trailers (0.360), 35 Other transport equipment (0.328), 36 Furniture; manufacturing n.e.c. (0.235), 37 Other manufactured products and recycling (0.339). Countries covered (102). Advanced Economies (35): Australia, Austria, Belgium, Canada, Cyprus, Czechia, Denmark, Estonia, Finland, France, Germany, Greece, Hong Kong, Iceland, Ireland, Israel, Italy, Japan, South Korea, Latvia, Lithuania, Luxembourg, Netherlands, New Zealand, Norway, Portugal, Singapore, Slovakia, Slovenia, Spain, Sweden, Switzerland, Taiwan, UK, USA. Emerging markets and developing economies (67): Algeria, Albania, Argentina, Armenia, Bangladesh, Barbados, Belarus, Bosnia, Brazil, Bulgaria, Burundi, Cameroon, Chile, China, Colombia, Costa Rica, Croatia, Dominican Republic, Ecuador, Egypt, Fiji, Honduras, Hungary, India, Indonesia, Iraq, Jamaica, Jordan, Kuwait, Kyrgyzstan, Lao, Lebanon, Lesotho, Macao, Malaysia, North Macedonia, Madagascar, Mauritius, Mexico, Moldova, Mongolia, Morocco, Myanmar, Nepal, Oman, Panama, Pakistan, Paraguay, Peru, Poland, Philippines, Romania, Russian Federation, Rwanda, Senegal, Serbia, Sri Lanka, South Africa, Thailand, Trinidad and Tobago, Tunisia, Turkey, Uruguay, Ukraine, Uzbekistan, Vietnam, Zambia. Distribution of the share of manufacturing value-added in GDP across countries in 2019 (in %): $Share_{i,c,t}$ min p10 p25 p50 p75 p90 max min p10 p25 p50 p75 p90 Total manufacturing over GDP Largest national industry over GDP 12.7 17.6 21.4 47.6 0.2All countries 0.6 6.3 9.51.3 2.0 2.8 3.9 5.1 18.0 **AEs** 1.2 5.8 9.612.4 18.7 22.6 31.5 0.41.2 1.8 2.1 3.2 5.1 14.6 **EMDEs** 0.6 6.3 8.9 12.6 17.2 19.9 47.6 0.21.4 2.4 3.0 5.1 18.0 Individual industries value-added as a share of GDP in 2019 and correlation of the individual industries' real growth with real GDP growth during 1980-2019 (in %): $Share_{i,c,t}$ min p10 p25 p50 p75 p90 p95 p10 p25 p50 p75 p90 p95 p99Individual manufactures over GDP Correlation with GDP growth All countries 0.03 0.1 0.30.71.5 2.4 4.6 -95.7 -11.4 15.4 41.9 63.8 77.6 82.9 **AEs** 0.040.10.40.8 2.1 3.9 32.4 49.8 67.3 78.7 83.3 1.5 -65.2 10.1

The paper uses as a measure of financial frictions, the external finance dependence index (EFD) of Rajan and Zingales. This index is defined as the ratio of capital expenditures that cannot be financed with cash-flow: (capital expenditures-cash-flow)/capital expenditures. The EFD is calculated as the median between 1980 and 1990 for the US firms in each industry available in Compustat. Note that the median and average external finance dependence across the 23 industries

EMDEs

 $0.03 \quad 0.1$

0.3

0.7

1.5

2.5

4.6

-95.7 -22.5

4.9 32.9 60.5 76.1 82.6

is 0.28 and 0.34, respectively. Furthermore, the interdecile range (i.e., the percentiles 10 to 90) of the external finance dependence of the industries goes from 0.03 to 0.95. Therefore, most industries have some degree of dependence of external funds, although two industries could be said to be externally finance independent (that is, with $EFD \leq 0$).

2.2 Empirical approach

The empirical approach exploits the differential behavior of each industry in the same sample of countries across time periods with banking crises or not. I estimate a panel data model of the real growth rate, $g_{i,c,t}$, for the industry i in country c at the time t, with an additive-linear form:

1)
$$g_{i,c,t} = \gamma EFD_i \times BankCrisis_{c,t} + \alpha Share_{i,c,t-1} + \delta x_{c,t} + f_{i,c} + f_t + \varepsilon_{i,c,t}$$
,

with $BankCrisis_{c,t}$ being a dummy denoting a banking crisis in country c at time t, EFD_i the external financial dependence of industry i, $Share_{i,c,t-1}$ is the fraction of value-added of industry i in the total manufacturing of the country in the previous year, $f_{i,c}$ is industry-country fixed effects, $f_{i,c}$ time fixed effects, $f_{i,c}$ is a vector of additional time-varying controls and $f_{i,c,t}$ is an idiosyncratic unobservable term. The estimation will be performed using the Correia 2017 estimator for models with a large number of fixed effects, $f_{i,c}$ using industry-country clusters and time dummies. The vector of time-varying controls $f_{i,c,t}$ includes variables such as $f_{i,c,t}$ and the log of the income per capita $f_{i,c,t}$ includes variables such as $f_{i,c,t}$ and the log of the income per capita $f_{i,c,t}$ includes variables such as $f_{i,c,t}$ and the log of the income per capita $f_{i,c,t}$ includes variables such as $f_{i,c,t}$ and $f_{i,c,t}$ and $f_{i,c,t}$ includes variables, which can attenuate the effects of financial crises (Madeira and Madeira 2019, Madeira et al. 2023).

Results also remain similar after controlling for the real GDP growth $(g_{c,t}^{GDP})$ and its lag $(g_{c,t-1}^{GDP})$. But in this case

⁴This variable is used in previous studies to account for larger and more developed industries growing gradually less over time (Braun and Larrain 2005, Kroszner et al. 2007, Cowan and Raddatz 2013).

⁵The standard Stata commands xtreg or reg with industry-country and time fixed effects give the same coefficient estimates and almost exactly the same standard errors as the reghtfe Correia command. The reg command sometimes gives small differences in the standard errors, even if the coefficients are the same.

⁶Note that in an online appendix I show that the results remain similar if one adds controls such as the inflation rate $(inflation_{c,t})$ and the monetary policy rate $(MPR_{c,t})$, with values for the current year (t) and the previous one (t-1). The inclusion of inflation and monetary policy do not change much the estimated models, except in the case of currency crises. Since currency crises often coincide with a surge in inflation, then the direct effect of the currency crises in the regression becomes smaller and part of it is captured as an inflation effect. The overall results, however, remain quite similar and the coefficients do not change much after inflation and monetary policy are controlled for.

Furthermore, I consider models with interactions such as time-changing effects:

2)
$$g_{i,c,t} = \gamma EFD_i \times BankCrisis_{c,t} \times S_{c,t} + \alpha Share_{i,c,t-1} + \delta x_{c,t} + f_{i,c} + f_t + \varepsilon_{i,c,t}$$
,

with $S_{c,t}$ being a dummy for a specific period such as the GFC.

Since the industries are heterogeneous in terms of external finance dependence, then the effect of financial crises is heterogeneous across industries and therefore the coefficient for the interaction $\gamma EFD_i \times BankCrisis_{c,t}$ is identified. Therefore, I consider additional robustness checks that include industry-country and country-year dummies (equation 3) and industry-country, industry-time and country-year fixed effects (equation 4):

3)
$$g_{i,c,t} = \gamma EFD_i \times BankCrisis_{c,t} \times S_{c,t} + \alpha Share_{i,c,t-1} + f_{i,c} + f_{c,t} + \varepsilon_{i,c,t}$$
,

4)
$$g_{i,c,t} = \gamma EFD_i \times BankCrisis_{c,t} \times S_{c,t} + \alpha Share_{i,c,t-1} + f_{i,c} + f_{i,t} + f_{c,t} + \varepsilon_{i,c,t}$$
.

One interesting issue that this paper studies is the effect of financial regulations taken after the GFC, which is here specified as the time period between 2007 and 2009. Therefore, I use as an additional control the macroprudential policy index from the iMaPP (integrated Macroprudential Policy) database published by the IMF (Alam et al. 2019). This database aggregates information on the countries' prudential policies from the IMF, BIS, FSB and national authorities. The iMaPP dataset has a set of 17 macroprudential categorical indicators (+1,0-1, for tightening, no change and easing, respectively). The 17 prudential policies include: Loan-to-value (LTV), Debt Service to Income (DSTI), Limits on Credit Growth (LCG), Loan Loss Provisions (LLP), Loan restrictions (LoanR), Limits and penalties to the loan-to-deposit (LTD), Limits on foreign currency lending (LFC), Reserve Requirements (RR), Liquidity, Limits on foreign exchange exposure (LFX), Leverage limits or unweighted Leverage Ratio (LVR), Countercyclical buffers (CCB), Conservation buffer, Capital requirements, Tax measures, Systemically Important Financial Institutions (SIFI) risk mitigation measures, Other measures (such as stress testing, restrictions on profit distribution

the interpretation for the isolated coefficient dummy $BankCrisis_{c,t}$ changes. The reason is that $BankCrisis_{c,t}$ also affects $g_{c,t}^{GDP}$ and $g_{c,t-1}^{GDP}$. Therefore, adding the GDP growth rate as controls implies that the coefficient is estimating the effect of banking crises on manufacturing growth, after its effect on economic output is already accounted for. For this reason, the coefficient of $BankCrisis_{c,t}$ on manufacturing growth tends to be small and insignificant. However, the coefficients for $EFD_i \times BankCrisis_{c,t}$ remain similar after accounting for ral GDP growth and its lag.

and limits on exposures between financial institutions). The iMaPP data reports a Total Prudential Policy index $(TPI_{c,t})$, which is the sum of the 17 macroprudential policies for a given period.

Table 2: Main industry and country variables used in the article and their sources

Variable	Description	Source
$g_{i,c,t}$	Growth rate of the real production index of the manufacturing	UNIDO
	industry i in country c in year t	
$ShareManVA_{i,c,t}$	Value-added of manufacture i as a fraction of the total	UNIDO
	manufacturing value-added of country c in year t	
$GDP_{c,t}^{PPP,pc}$	GDP per capita of country c in year t (2017 USD in PPP)	World Bank
$BankCrisis_{c,t}$	Dummy with value 1 if country c in year t is facing a banking	Laeven &
	crisis, 0 otherwise (Laeven and Valencia 2020, Nguyen et al 2022)	Valencia
$CurrencyCrisis_{c,t}$	Dummy with value 1 if country c in year t is facing a currency	Laeven &
	crisis, 0 otherwise (Laeven and Valencia 2020, Nguyen et al 2022)	Valencia
$DebtCrisis_{c,t}$	Dummy with value 1 if country c in year t is facing a sovereign debt	Laeven &
	crisis, 0 otherwise (Laeven and Valencia 2020, Nguyen et al 2022)	Valencia
$AnyCrisis_{c,t}$	Dummy with value 1 if country c in year t is facing a financial	Laeven &
	crisis, 0 otherwise (Laeven and Valencia 2020, Nguyen et al 2022)	Valencia
Crisis	Output loss given by the cumulative sum of the difference between	Laeven &
$OutputLoss_{c,t}$	actual and trend real GDP over the period [T, T+3], in % of GDP	Valencia
$TPI_{c,t}$	Total prudential policy index: net sum of all macroprudential	iMaPP
	tightening measures $(+1,0,-1)$ taken by country c in year t (including	(IMF)
	17 different types of regulation affecting borrowers or lenders)	
$CTPI_{c,t}^{2010}$	Cumulative total prudential policy index: net sum of all prudential	iMaPP
,	tightening measures in country c since 2010 until year t	(IMF)
EFD_i	External Finance Dependence index of manufacture i : the fraction	Rajan &
	of capital expenditures of the median firm that is not financed with	Zingales
	cash-flow: (capital expenditures-cash-flow)/capital expenditures.	(1998)
	Calculation for the US firms in Compustat between 1980 and 1989.	,

As shown in previous research (Akinci and Olmstead-Rumsey 2018, Alam et al. 2019), macroprudential policies started being increasingly used by most countries after the GFC. Therefore, I consider macroprudential policies implemented after the GFC by including as additional controls the variables for the macroprudential policy implemented in the current year, $TPI_{c,t} \times 1(t \ge 2010)$, and the cumulative macroprudential policy stance of all measures implemented since 2010, $CTPI_{c,t-1}^{2010} = \sum_{t^*=t_0}^{t-1} TPI_{c,t^*} \times 1(t^* \ge 2010)$, with t_0 denoting the beginning year of the dataset.

Table 2 summarizes all the variables used in this article and their sources. All the data used in this article are publicly available. Note that the dummies for banking, currency, sovereign debt and any financial crisis in this article come from Laeven and Valencia 2020 for the period between

1980 to 2017, but these are updated until 2019 with data from Nguyen et al. 2022. Furthermore, all codes required to replicate the analysis will be available in a Mendeley Data link.

3 Main results for banking crises

The main model estimates are shown in Table 3. The results show that all industries (even those with no external finance dependence, i.e., with $EFD \leq 0$) are affected by banking crises, at least in the samples for all countries, AEs and EMDEs. Even an industry with no external dependence experiences a decline in growth of 2.4%, 2.5% and 0.8% in the samples of all countries, AEs and EMDEs. This effect is statistically significant, besides being economically sizeable. The regressions across all the country groups show that externally dependent industries are much more affected during banking crises, especially in EMDEs.

Externally dependent industries (i.e, with EFD > 0) experience a further negative growth effect during banking crises, confirming the results of Dell'Ariccia et al. 2008. The coefficient of $EFD_i \times BankCrisis_{c,t}$ for all countries is estimated to be -3.39 and -3.56 in the regressions with macroeconomy controls and country-year fixed effects. This effect is somewhat higher in absolute value that the -2.74 coefficient estimated by Dell'Ariccia et al. 2008 for a sample of 41 countries during the period between 1980 to 2000. However, these coefficients are not statistically different from the Dell'Ariccia et al. 2008 estimates if a two standard deviation interval is considered.

Externally dependent industries in EMDEs experience reductions in growth that are more than twice as large. Fully externally dependent industries (that is, industries with EFD = 1) during banking crises experience a reduction in growth of 3.4%, 0.9% and 7.5% in the samples for all countries, AEs and EMDEs, respectively, using the regressions with macroeconomic controls. In the regressions with country-year fixed effects, fully externally dependent industries during banking crises experience a reduction in growth of 3.6%, 1.2% and 7.2% in the samples for all countries, AEs and EMDEs, respectively. Note that only the coefficients for the advanced economies are statistically insignificant. The all countries and emerging markets coefficients are significant both in the specifications with macroeconomic controls and with country-year fixed effects.

Finally, the results in Table 3 show that industrial growth is declining relative to the development of the economy as given by its GDP per capita and relative to the size of the individual industry in terms of the total manufacturing. Both results make sense. As countries develop and reach the technological frontier, its manufacturing growth falls. It is likely that individual industries present lower growth as their size becomes big relative to their national economy and their resource demands for labor and materials become more expensive. These two results are statistically significant and quite similar in all the models estimated in this article (see also Madeira 2024).

Table 3: Effects of banking crises on industrial growth across country groups

	O		_	,	•	0 1
Controls	With mac	croeconomi	c controls	With cour	ntry-year fi	ixed effects
	All	Advanced	EMDEs	All	Advanced	EMDEs
	countries	economies		countries	economies	
$EFD_i \times$	-3.389***	-0.824	-7.502***	-3.562***	-1.153	-7.188***
$BankCrisis_{c,t}$	(1.075)	(1.324)	(1.738)	(0.963)	(1.224)	(1.466)
$ShareManVA_{i,c,t-1}$	-0.164***	-0.149***	-0.170***	-0.154***	-0.136***	-0.162***
	(0.0359)	(0.0427)	(0.0484)	(0.0331)	(0.0409)	(0.0454)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.882***	-7.635***	-6.346***			
,	(0.597)	(0.820)	(1.081)			
$BankCrisis_{c,t}$	-2.398***	-2.457***	-0.791			
	(0.394)	(0.504)	(0.637)			
N	41,648	20,228	21,420	41,980	20,472	21,508
R^2 (overall)	0.142	0.242	0.107	0.315	0.369	0.292

Robust standard errors in (). Clusters by industry-country. ***,**,* denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

Now I show how the effects of banking crises differed during the Great Financial Crisis (hence on GFC), defined as a time dummy between 2007-2009. Table 4 shows that the effects of banking crises on all types of industries were particularly strong during the GFC, especially in EMDEs. Banking crises during the GFC had a further reduction in growth of 2.7% and 9.6% in the samples of all countries and EMDEs, in addition to the usual negative effect of banking crises. For the AEs the effect is more complicated to determine, since the coefficient for the interaction of banking crises with the GFC period was positive for the effect of all industries, but with a similarly sized coefficient for the externally dependent industries. The regressions with country-industry and country-year fixed effects do not show an additional effect of the GFC on externally dependent industries for any of the country samples (all countries, AEs, EMDEs). These regressions show estimates similar to Table 3, indicating a reduction in growth of fully dependent industries during banking crises of 3.9%, 0.7% and 7.3% for all countries, AEs and EMDEs.

Table 4: Effects of banking crises on industrial growth, with interactions for the Great Financial Crisis (2007-2009)

	Great	i ilialiciai	C11616 (200	. 2000)		
Controls	With mac	croeconomi	c controls	With cour	ntry-year fi	xed effects
	All	Advanced	EMDEs	All	${\bf Advanced}$	EMDEs
	countries	economies		${\rm countries}$	economies	
$EFD_i \times Bank$	-3.474***	0.193	-7.546***	-3.962***	-0.730	-7.288***
$Crisis_{c,t}$	(1.277)	(1.817)	(1.683)	(1.156)	(1.694)	(1.494)
$EFD_i \times Bank$	0.714	-2.613	1.388	1.505	-1.068	1.443
$Crisis_{c,t} \times GFC_t$	(1.992)	(2.137)	(8.284)	(1.813)	(1.980)	(8.164)
$ShareManVA_{i,c,t-1}$	-0.164***	-0.149***	-0.170***	-0.154***	-0.136***	-0.162***
	(0.0359)	(0.0429)	(0.0484)	(0.0331)	(0.0409)	(0.0454)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.824***	-7.638***	-6.214***			
	(0.597)	(0.821)	(1.072)			
$BankCrisis_{c,t}$	-1.847***	-3.304***	-0.0809			
	(0.448)	(0.637)	(0.611)			
$BankCrisis_{c,t} \times$	-2.687**	2.755**	-9.593**			
GFC_t	(1.086)	(1.177)	(3.999)			
N	41,648	20,228	21,420	41,980	20,472	21,508
R^2 (overall)	0.142	0.243	0.108	0.315	0.369	0.292
D 1 4	. 1 1	• ()	C1 / 1			

Robust standard errors in (). Clusters by industry-country.

***,**,* denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

4 Effects of other types of financial crises on industrial growth

Now I analyze the effects of other financial crises, including currency and sovereign debt crises. Note that banking and currency crises are much more common than sovereign debt crises (Laeven and Valencia 2020), therefore the latter have fewer observations and more imprecise results. Currency crises often precede or coincide with banking crises. One potential channel for this is the negative balance sheet effect for companies borrowing from abroad, which see their debts appreciate during currency crises. Firms can then become delinquent and transmit their losses to the financial system.

Table 5 shows the effects on industrial growth of currency, sovereign debt and any type of financial crises in separate regressions and in a joint regressions with all types of crises. The models are estimated with both macroeconomic control variables and country-year fixed effects. In the regressions with macro controls there are two effects of financial crises, one a direct effect on all industries and then an additional effect on externally dependent industries.

Currency crises have an even larger negative effect on externally dependent industries than banking crises, whether in terms of the direct effect on all industries or in terms of the additional effect on externally dependent industries. Sovereign debt crises also have strong negative effect on all industries and on externally dependent industries' growth, but with less impact than banking and currency crises. The regressions with all types of crises in the same model confirm these results. However, the effect of sovereign debt crises on externally dependent industries is not statistically significant in the regressions with all types of crises. All the results are robust to considering either macroeconomic controls or country-year fixed effects.

Table 5: Effects of different types of financial crises (banking, currency, sovereign debt, any financial crisis), across the entire country sample

Controls	With	macroeco		$\frac{1}{1}$	• -	Vith count	ry-year fixed	d effects
$EFD_i \times Banking$				-2.910***				-2.993***
$Crisis_{c,t}$				(1.092)				(0.990)
$EFD_i \times Currency$	-5.086***			-3.932**	-5.048***			-4.034**
$Crisis_{c,t}$	(1.764)			(1.828)	(1.794)			(1.871)
$EFD_i \times Sovereign$		-2.858**		-1.736		-2.135*		-1.076
$DebtCrisis_{c,t}$		(1.366)		(1.333)		(1.126)		(1.155)
$EFD_i \times Any$			-2.906***				-2.832***	
$Crisis_{c,t}$			(0.901)				(0.821)	
$ShareManVA_{i,c,t-1}$	-0.163***	-0.164***	-0.168***	-0.165***	-0.152***	-0.153***	-0.154***	-0.154***
	(0.0357)	(0.0357)	(0.0360)	(0.0359)	(0.0330)	(0.0329)	(0.0331)	(0.0332)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.123***	-6.797***	-7.062***	-6.483***				
5,0 1	(0.583)	(0.597)	(0.603)	(0.599)				
$BankCrisis_{c,t}$,	,	,	-1.774***				
,				(0.405)				
$CurrencyCrisis_{c,t}$	-5.247***			-4.794***				
,	(0.788)			(0.818)				
$DebtCrisis_{c,t}$		-1.207**		-0.439				
,		(0.478)		(0.488)				
$AnyCrisis_{c,t}$,	-2.375***	, ,				
,			(0.360)					
N	41,648	41,648	41,648	41,648	41,980	41,980	41,980	41,980
R^2 (overall)	0.145	0.139	0.143	0.147	0.315	0.315	0.315	0.316

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

Summarizing the regressions with all financial crises included, the results with macroeconomic controls show that the effect on all industries is a reduction in growth of 1.8%, 4.8% and 0.4% for

banking, currency and sovereign debt crises, respectively. Fully externally dependent industries experience a further reduction in growth of 2.9%, 3.9% and 1.7% during banking, currency and sovereign debt crises, respectively. Therefore, currency crises have a much stronger effect on industrial growth whether directly on all industries or as an additional effect through their external finance dependence. Note that the coefficients for sovereign debt crises are statistically insignificant in the regressions with all types of crises, although being significant in the separate regressions.

I then estimate effects of all types of financial crises across country groups (AEs, EMDEs). In terms of the direct effect of crises on all industries, Table 6 shows that banking, currency and debt crises reduce growth in AEs by 2.1%, 5.6% and 2%, respectively. For EMDEs the direct effect of banking and sovereign debt crises is small and statistically insignificant, but currency crises reduce the growth of all industries by 4.7%.

Table 6: Effects of different types of financial crises across country groups

		•		across country groups
Controls	With mac	ro controls	With cour	ntry-year fixed effects
	Advanced	EMDEs	Advanced	EMDEs
	economies		economies	
$EFD_i \times Banking$	-1.047	-6.266***	-1.137	-6.182***
$Crisis_{c,t}$	(1.300)	(1.830)	(1.217)	(1.560)
$EFD_i \times Currency$	4.721	-4.418**	1.075	-4.215**
$Crisis_{c,t}$	(3.682)	(2.026)	(4.709)	(2.074)
$EFD_i \times Sovereign$	-4.281	-0.672	-4.035	-0.287
$DebtCrisis_{c,t}$				(1.202)
$ShareManVA_{i,c,t-1}$	-0.150***	-0.170***	-0.137***	-0.161***
			(0.0410)	(0.0455)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-7.760***	-5.747***		
,	(0.825)	(1.080)		
$BankCrisis_{c,t}$	-2.143***	-0.145		
	(0.496)	(0.662)		
$CurrencyCrisis_{c,t}$	-5.661***	-4.738***		
	(1.768)	(0.902)		
$DebtCrisis_{c,t}$	-1.957	-0.129		
	(1.415)	(0.547)		
N	20,228	21,420	20,472	21,508
R^2 (overall)	0.244	0.113	0.369	0.292

Robust standard errors in (). Clusters by industry-country.

***,**,* denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

The effects of financial crises on externally dependent industries in AEs are not statistically

significant. However, for EMDEs banking and currency crises imply an additional growth reduction of 6.3% and 4.4% for fully dependent industries. These results are confirmed by either the regressions with macroeconomic controls or country-year fixed effects, with similar coefficients for both models. In the case of the regressions with macroeconomic controls for EMDEs, there is a stronger effect of currency crises on all industries, although banking crises have a stronger impact through the external finance dependence channel.

As a robustness check, I estimate a quantile regression model with country-industry fixed effects, using the methodology proposed by Machado and Santos-Silva 2019. This methodology allows us to test the effect of financial crises effects under different periods of each industry's growth, whether a low (say, quantile 25), median (quantile 50) or high growth (quantile 75) period. Quantile regression is also robust to outliers, such as a few extreme observations (Machado and Santos-Silva 2019).

The regressions in Table 7 show a strong direct effect of banking and currency crises on all industries in the cases of low and median growth periods (quantiles 25 and 50). Currency crises also have a strong direct effect on all industries in periods of high growth (quantile 75). Banking crises reduce the growth of all industries by 4.1%, 2.5% and 1.1% at the quantiles 25, 50 and 75. Currency crises reduce the growth of all industries by 6.5%, 4.7% and 3.1% at the quantiles 25, 50 and 75. It is also noticeable that the coefficients for currency crises are statistically significant for all the quantiles. For banking crises there is statistically significance for the quantiles 25 and 50. The effects estimated for sovereign debt crises in the all countries sample are small and statistically insignificant for all quantiles.

For the country groups (AEs, EMDEs) only the median quantile regression is reported. The median quantile shows a strong negative effect of banking and sovereign debt crises on all industries in AEs. The financial crises coefficients are not statistically significant for the EMDEs, although the coefficient size is large in the case of currency crises,

The results in Table 7 do not show statistically significant coefficient for the effect of financial crises on externally dependent industries. However, the size and sign of the coefficients confirm the previous results in this article. Banking, currency and sovereign debt crises have a negative effect on the externally dependent industries, particularly in periods of low and median growth. Banking and currency crises seem to have a much stronger effect on EMDEs. Sovereign debt crises only have a sizeable effect on externally dependent industries in AEs. However, none of these coefficients are

statistically significant in the quantile regressions. This makes sense due to the higher standard errors involved in the moment conditions of panel quantile regressions (Machado and Santos Silva 2019).

Table 7: Quantile regressions with country-industry fixed effects for the impact of different types of financial crises (banking, currency, sovereign debt)

of different type	s of financ	iai crises (banking, o	currency, so	overeign debt)
	A	ll countrie	es	Advanced	EMDEs
				economies	
Controls	Q25	Q50	Q75	Q50	Q50
$EFD_i \times Banking$	-3.932	-3.008	-2.151	-1.245	-5.901
$Crisis_{c,t}$	(3.381)	(2.154)	(2.083)	(1.604)	(71.09)
$EFD_i \times Currency$	-3.769	-3.765	-3.762	4.609	-4.560
$Crisis_{c,t}$	(5.431)	(3.460)	(3.345)	(5.142)	(68.67)
$EFD_i \times Sovereign$	-3.550	-1.782	-0.141	-4.207	-0.754
$DebtCrisis_{c,t}$	(4.131)	(2.632)	(2.545)	(6.150)	(50.90)
$ShareManVA_{i,c,t-1}$	-0.0673	-0.173***	-0.271***	-0.171***	-0.173
	(0.104)	(0.0661)	(0.0639)	(0.0418)	(1.673)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-4.117***	-6.859***	-9.403***	-7.631***	-5.485
	(1.470)	(0.937)	(0.906)	(0.618)	(31.86)
$BankCrisis_{c,t}$	-4.070***	-2.548***	-1.137	-3.350***	-1.039
	(1.304)	(0.831)	(0.804)	(0.621)	(26.16)
$CurrencyCrisis_{c,t}$	-6.474***	-4.724***	-3.101**	-3.472	-5.044
	(2.014)	(1.284)	(1.241)	(2.290)	(25.09)
$DebtCrisis_{c,t}$	-0.279	-0.672	-1.037	-3.706**	-0.389
	(1.532)	(0.976)	(0.944)	(1.802)	(19.47)
$GreatModeration_t$	-1.253*	-1.079**	-0.918**	-1.476***	-0.687
(1988-2006)	(0.664)	(0.423)	(0.409)	(0.259)	(16.17)
GFC_t	-7.033***	-4.433***	-2.021***	-5.944***	-3.100
(2007-2009)	(0.942)	(0.600)	(0.581)	(0.470)	(17.29)
N	41,648	41,648	41,648	20,228	21,420

Standard errors in ().

***, **, * denote 1%, 5%, 10% statistical significance.
All regressions include fixed effects by industry-country (omitted).

Finally, Table 8 reports the effects of the financial crises across all country groups (all countries, AEs and EMDEs), with a full set of country-industry, country-year and industry-year fixed effects. These results show that the estimated effects of financial crises on externally dependent industries are robust. For the regressions with all types of financial crises, the sample of all countries shows that fully externally dependent industries experience a reduction in growth of 2.6% and 4.1% during banking and currency crises. However, across country groups, the only statistically significant

effects are found for the EMDEs. Fully financially dependent industries in EMDEs experience a growth reduction of 6.3% and 4.9% during banking and currency crises. Sovereign debt crises are not statistically significant for any sample in the all crises model. However, it is relevant to note that all types of financial crises have relevant and statistically significant effects in the models with separate regressions. This is the case for both the samples of all countries and EMDEs. Interestingly, both in the models with all financial crises in the same regression and in separate regressions, the estimated effect on externally dependent industries is always stronger for the EMDEs.

Table 8: Effects of different types of financial crises with industry-country, country-year and industry-year fixed effects

All	Advanced	EMDEs
countries	economies	
in the sar	ne regression	on model
-2.612***	-1.479	-6.290***
(1.011)	(1.618)	(1.554)
-4.133**	1.381	-4.847**
(1.964)	(4.340)	(2.317)
-0.972	-0.693	-1.300
(1.107)	(3.720)	(1.264)
-0.158***	-0.158***	-0.159***
(0.0341)	(0.0503)	(0.0449)
41,910	20,400	21,480
0.343	0.426	0.330
	in the sar -2.612*** (1.011) -4.133** (1.964) -0.972 (1.107) -0.158*** (0.0341) 41,910	countries economies in the same regressic -2.612*** -1.479 (1.011) (1.618) -4.133** 1.381 (1.964) (4.340) -0.972 -0.693 (1.107) (3.720) -0.158*** -0.158*** (0.0341) (0.0503) 41,910 20,400

Coefficients obtained with different types of crises in separate

regressions									
$EFD_i \times Banking$	-3.183***	-1.425	-7.221***						
$Crisis_{c,t}$	(0.986)	(1.579)	(1.535)						
$EFD_i \times Currency$	-4.944***	0.597	-6.054***						
$Crisis_{c,t}$	(1.894)	(4.244)	(2.258)						
$EFD_i \times Sovereign$	-1.907*	-1.012	-2.328*						
$DebtCrisis_{c,t}$	(1.094)	(3.679)	(1.273)						
$EFD_i \times Any$	-2.533***	-1.592	-5.503***						
$Crisis_{c,t}$	(0.867)	(1.439)	(1.537)						
N	41,910	20,400	21,480						

All regressions include $ShareManVA_{i,c,t-1}$ as a control, but its coefficient is omitted in the regressions with each type of crisis separately.

Clusters by industry-country and industry-year. Robust standard errors in ().

***, **, * denote 1%, 5%, 10% statistical significance. Fixed effects in the regressions are omitted. For the separate regressions with each type of crisis, the results show a statistical significant effect of all types of crises on the samples of all countries and EMs. Fully dependent industries experience a growth reduction of 3.2%, 4.9% and 1.9% for banking, currency and sovereign debt crises in the all countries sample. In EMs the effect on externally dependent industries is much stronger for banking crises. Fully dependent industries experience a growth reduction of 8.8%, 4.1% and 3% for banking, currency and sovereign debt crises in EMs. These estimates confirm the previous results, showing stronger effects on externally dependent industries in EMs during financial crises. Currency crises have a large effect on LICs, but the effect is not statistically significant (similarly to previous tables). The estimated result of banking crisis effects just for EMDEs fits well with Dell'Ariccia et al. 2008, who also find only a negative impact of banking crises on the industrial growth of externally dependent industries for developing countries. However, the coefficient of -7.22 shown in Table 8 for EMDEs is almost twice the size of the -3.73 value reported in Dell'Ariccia et al. 2008.

5 Total impact of financial crises on manufacturing growth

The previous sections show the effects on growth of a particular industry. Coefficients must be interpreted as giving an effect on industries with no external finance dependence (EFD=0) or an additional growth effect on industries with a specific external finance dependence value. However, industries have fairly different levels of EFD, as shown in Table 1. Furthermore, countries have different industrial specializations and their industries change over time. Therefore, the EFD of the total manufacturing sector differs both across countries and time.

For this reason, Table 9 summarizes the effect on total manufacturing implicit in the previous exercises. In the case of the additional growth effect experienced by industries with external finance dependence, the total effect on the manufacturing sector of a country c at time t can be obtained as: $\theta_{c,t} = \gamma \sum_i \varpi_{i,c,t} EFD_i$, with the weight $\varpi_{i,c,t} = \frac{VA_{i,c,t}}{\sum_i VA_{i,c,t}}$ being the share of each individual industry in terms of the value added of the entire national manufacturing sector. In the case of the regressions that include country-year fixed effects, this is the only term being estimated. For the regressions with macroeconomic controls, there is also the effect of financial crises on all industries (even those with EFD = 0). Therefore, in this case the total effect on the manufacturing sector

is given by: $\mu_{c,t} = \delta_{Crisis} + \gamma \sum_i \varpi_{i,c,t} EFD_i$. Table 9 reports both kinds of effects. Furthermore, since the effects differ across countries, Table 9 shows the average effects on the different country samples, that is: $\frac{1}{n_S} \sum_{c \in S} \theta_{c,t}$ and $\frac{1}{n_S} \sum_{c \in S} \mu_{c,t}$, with S denoting either the sample of all countries, AEs or EMDEs, and n_S the number of countries in each sample.

Table 9: Effect on total manufacturing growth (in %) of different types of financial crises (average across countries in 2019): estimates from all the models

financial crises (average across countries in 2019): estimates from all the models									
All crises in the same regression model									
Coefficient average	Triple	FE (i,c+i	i,t+c,t)	Doub	le FE (i,c	+c,t)	FE (i,c+t) plus	controls
across countries	$\gamma \sum$	$\sum_{i} \varpi_{i,c,t} E_{i}$	FD_i	$\gamma \sum$	$\sum_{i} \varpi_{i,c,t} E I$	FD_i	δ_{Crisi}	$s + \gamma \sum_{i} \varpi$	$_{i,c,t}EFD_{i}$
Crisis type	Bank	Currency	Debt	Bank	Currency	Debt	Bank	Currency	Debt
All countries	-0.8	-1.3	-0.3	-1.0	-1.3	-0.3	-2.7	-6.0	-1.0
AEs	-0.5	0.5	-0.2	-0.4	0.4	-1.3	-2.5	-4.1	-3.4
EMDEs	-2.0	-1.5	-0.4	-2.0	-1.3	-0.1	-2.1	-6.1	-0.3
		Separate	e mode	ls for e	ach crisis	type			
Coefficient average	Tripe	e FE (i,c i	, t c, t)	Doub	le FE (i,c	+c,t)	FE (i,c+t) plus	controls
across countries	$\gamma \sum$	$\sum_{i} \varpi_{i,c,t} E_{i}$	FD_i	$\gamma \sum$	$\sum_{i} \varpi_{i,c,t} E I$	FD_i	δ_{Crisi}	$s + \gamma \sum_{i} \varpi$	$_{i,c,t}EFD_{i}$
Crisis type	Bank	Currency	Debt	Bank	Currency	Debt	Bank	Currency	Debt
All countries	-1.0	-1.6	-0.6	-1.1	-1.6	-0.7	-3.5	-6.9	-2.1
AEs	-0.5	0.2	-0.3	-0.4			-2.7		
EMDEs	-2.3	-1.9	-0.7	-2.3			-3.2		

Note: FE denotes the set of fixed effects, whether a simple pair (i,c+t: industry-country and time), a double set of pairs (i,c+c,t: industry-country and country-time) or a triple set of pairs (i,c+i,t+c,t: industry-country, industry-time and country-time).

Table 9 summarizes the total industry impact obtained from the triple fixed effect regressions (Table 8), the double fixed effect regressions (Tables 3, 5 and 6) and the fixed effects model with macroeconomic controls (Tables 3, 5 and 6). For simplicity, I show only the effects for the last year of the sample, which corresponds to the last pre-pandemic year, 2019. The results for the triple and the double fixed effect models are almost the same, both qualitatively and in size. The results are also similar, whether qualitative or in size, for the simultaneous regression with all crises and the separate regressions. Therefore, I comment only on the results from the simultaneous regressions.

The coefficients on total manufacturing due to the external finance dependence of industries $(\gamma \sum_i \varpi_{i,c,t} EFD_i)$ show that EMDEs are much more affected by banking crises. The triple FE estimates show that the average country experiences a manufacturing growth loss of 0.8%, 13% and 0.3% during banking, currency and sovereign debt crises, respectively. For AEs, there is a reduction in manufacturing growth of 0.5% and 0.2% during banking and sovereign debt crises,

while currency crises have no effect. For EMDEs there is a reduction in growth of 2%, 1.5% and 0.4% during banking, currency and sovereign debt crises.

Considering the total effect from the model with fixed effects and macroeconomic controls, it is shown that currency crises have the strongest impact across all country groups (all countries, AEs, EMDEs). There is a loss in industrial growth of 2.7%, 6% and 1% during banking, currency and sovereign debt crises in the all countries sample. EMDEs experience the strongest effects from currency crises, while AEs experience the strongest effect of banking and sovereign debt crises. In AEs there is a reduction in industrial growth of 2.5%, 4.1% and 3.4% during banking, currency and debt crises. In EMDEs there is a loss of industrial growth of 2.1%, 6.1% and 0.3% during banking, currency and sovereign debt crises. EMDEs are the countries least affected by banking crises, which could be due to the low development of their banking sector (Rajan and Zingales 1998, Raddatz 2006). Perhaps the stronger effect of sovereign debt crises in AEs is due to their larger sovereign debt markets and the unexpected of sovereign crises in advanced economies. In EMDEs the level of sovereign debt relative to the economy is much smaller, which could explain the small effect of sovereign debt crises on these countries.

Table 10: Effects on total manufacturing growth (in %, yearly) of banking crises compared with Dell'Ariccia et al. 2008 sample: triple FE (i,c+i,t+c,t)

crises compared with Den Affecta et al. 2008 sample. The (i,c $+$ i,t $+$ c,t)							
	All crises in	same regression	Crises types in	separate regressions			
Source	This work	${\bf Dell'Ariccia\ et\ al.}$	This work	Dell'Ariccia et al.			
Period	1980 - 2019	1980-2000	1980-2019	1980-2000			
Country sample	102 countries	41 countries	102 countries	41 countries			
		Bank	ing crises				
All countries	-0.8	-0.3	-1.0	-0.7			
AEs	-0.5	0.6	-0.5	0.7			
EMDEs	-2.0	-1.0	-2.3	-1.3			
		Curre	ncy crises				
All countries	-1.3	-1.4	-1.6	-1.5			
AEs	0.5	0.1	0.2	0.7			
EMDEs	-1.5	-1.5	-1.9	-1.8			
		Sovereign	n debt crises				
All countries	-0.3	-0.2	-0.6	-0.4			
AEs	-0.2	0.0	-0.3	0.0			
EMDEs	-0.4	-0.2	-0.7	-0.5			

Note: Triple FE (i,c+i,t+c,t) denotes fixed-effects for industry-country, industry-time, country-time.

Finally, Table 10 summarizes the results of the triple fixed effect model (with country-industry,

country-year and industry-year dummies) in this article relative to the sample of Dell'Ariccia et al. 2008. The results show that the estimates in this article for currency and sovereign debt crises are similar to those implicitly estimated by Dell'Ariccia et al. 2008,⁷ whether for all countries, AEs and EMDEs. However, the calculations in this article show that my estimates of the effect of banking crises are much larger than those in previous work. Using the regressions for all types of financial crises in the same regression, I estimate a reduction in the growth of externally dependent industries of 0.8%, 0.5% and 2% for all countries, AEs and EMDEs, which is more than twice the effect from the Dell'Ariccia et al. 2008 sample. Furthermore, one cannot find negative effects of banking crises on AEs in the Dell'Ariccia et al. 2008 sample. Note that the time period of Dell'Ariccia et al. 2008 ends in 2000, therefore excluding the Great Financial Crisis. This could explain their lower estimate for the effect of banking crises, particularly in the case of AEs.

Furthermore, the Dell'Ariccia et al. 2008 study only estimated the effect of banking crises on the externally dependent industries, but did not include the direct effect on all industries that is estimated in this article. The total effect on manufacturing in this article is obtained from the regression with macroeconomic controls, country-industry and time fixed effects. Table 9 shows that including this direct effect for all countries makes the estimates of banking, currency and sovereign debt crises more than three times larger. For the all countries sample in Table 9, the total reduction in manufacturing growth is 2.7%, 6% and 1% for banking, currency and sovereign debt crises, which are estimates more than four times larger than the externally dependence of industries' effect obtained in the Dell'Ariccia et al. 2008 sample. The estimates in this article (in the last 3 columns of Table 9) for the costs of financial crises in terms of the total effect on manufacturing growth are nine, four and five times larger for banking, currency and sovereign debt crises relative to the Dell'Ariccia et al. 2008 sample.

⁷Note that Dell'Ariccia et al. 2008 only report the coefficients from their regressions. Their article does not report the total effect on the externally dependent industries, which requires weighting the impact according to the value-added of each industry. Therefore these calculations are obtained by replicating their analysis on their 41 country sample for the period between 1980 to 2000.

⁸Note that the sample of Dell'Ariccia et al. 2008 was only about one third of the sample used in this article. Dell'Ariccia et al. 2008 were using only about half of the sample in terms of time periods (1980-2000) and less than half in terms of countries (41 countries). Therefore, their smaller sample made it harder to estimate more coefficients and obtain statistically significant results. However, the estimates for the external finance channel of financial crises on manufacturing growth are quite similar between this article and the study of Dell'Ariccia et al. 2008.

6 Effects of financial policy during banking crises

Now I estimate a model with interaction components for the intensity of the banking crises, as expressed by its total output loss as a fraction of GDP (Laeven and Valencia 2020) and a dummy for the GFC. The dummy for the GFC in this regression takes into account both the time period between 2007 and 2009 and also crises that started in that period even if such crises finished afterwards, that is: $GFC_{c,t} = 1(\max(t, BegBankCrisis_{c,t}) \ge 2007)1(\min(t, BegBankCrisis_{c,t}) \le 2009)$, with $BegBankCrisis_{c,t}$ denoting the year in which the crisis began (Laeven and Valencia 2020). Furthermore, the model also considers the interaction effect of macroprudential policies introduced after the GFC, which were widely adopted across both developed and developing economies after 2010 (Akinci and Olmstead-Rumsey 2018, Alam et al. 2019). Therefore, the model considers the effect of macroprudential policies implemented after the GFC, measured as the net macroprudential policy tightening implemented in the current year, $TPI_{c,t} \times 1(t \ge 2010)$, and the cumulative macroprudential policy stance with all measures implemented since 2010, $CTPI_{c,t-1}^{2010}$. Note that the analysis in this section is limited to 2017 due to the availability of data on the output loss of banking crises (Laeven and Valencia 2020).

Table 11 shows the results of this model that accounts for both the intensity of the banking crises and the financial policies that have been implemented since 2010. The regressions with macroeconomic controls show that the output loss of banking crises has a negative effect on all industries for the samples of all countries and AEs. Macroprudential policies had a positive impact on industrial growth during the banking crises for the period after 2010 across all country samples (all countries, AEs, EMDEs), as seen by the coefficient for $BankCrisis_{c,t} \times CTPI_{c,t-1}^{2010}$.

For the coefficients that interact with the industries' external finance dependence, there is a negative effect of banking crises on externally dependent industries in the samples of all countries and EMDEs. The output loss of banking crises causes a negative effect on externally dependent industries in AEs.¹¹ I also find that externally dependent industries are negatively affected by

⁹Note that now the coefficients for $BankCrisis_{c,t}$ and $EFD_i \times BankCrisis_{c,t}$ no longer represent the full effect of a banking crisis on industries, because the model takes into account several interactions with other variables.

 $^{^{10}}$ It is worth noting, however, that the results do not change much if one uses instead the simple time period criterion for the GFC: $GFC_{c,t} = 1(2009 \ge t \ge 2007)$.

¹¹Note that the coefficient for the interaction between external finance dependence and the GFC period is now

the previous accumulated macroprudential tightening in the sample of all countries. This result makes sense, because these industries have more needs of external financial funds (Madeira 2024). Furthermore, macroprudential policy easing during banking crises (that is, the opposite sign of $EFD_i \times TPI_{c,t} \times 1(t \ge 2010)$) has a positive impact on growth of externally dependent industries for the samples of all countries and AEs.

positive for the externally dependent industries in all countries and AEs. However, this does not mean that the GFC was positive for growth, since the GFC implied a large loss in output and this output loss causes lower growth. The coefficient for the GFC effect on all industries is still negative, both in the sample of all countries and for EMDEs.

Table 11: Effects of banking crises on industrial growth, with controls for the intensity of the crisis and for post-GFC net macroprudential policy tightening

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		With mac	roeconomic	controls,	With cou	ntry-industr	ry, country-year
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $							
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Controls	All	Advanced	EMDEs	All	Advanced	EMDEs
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		countries	${\bf economies}$		countries	economies	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$EFD_i \times$	-5.291***	0.115	-8.413***	-6.248***	-0.232	-9.257***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$BankCrisis_{c,t}$	(1.894)	(3.359)	(2.336)	(1.544)	(3.077)	(2.286)
$EFD_{i} \times GFC_{c,t} 6.284^{***} 2.954 -1.320 9.162^{***} 5.209^{*} 4.230 \\ (2.149) (3.192) (8.257) (2.002) (3.110) (7.244) \\ EFD_{i} \times TPI_{c,t} -1.774 -3.508^{*} -0.0610 -2.358^{*} -3.997^{***} -0.994 \\ \times 1(t \geq 2010) (1.328) (2.017) (2.194) (1.371) (1.244) (2.426) \\ EFD_{i} \times CTPI_{c,t-1}^{2010} -2.597^{**} -1.442 -1.896 -2.229^{**} -0.222 -1.810 \\ (1.188) (1.471) (1.364) (1.038) (1.091) (1.375) \\ ShareManVA_{i,c,t-1} -0.210^{***} -0.187^{***} -0.214^{***} -0.190^{***} -0.190^{***} -0.189^{***} \\ (0.0442) (0.0491) (0.0621) (0.0433) (0.0648) (0.0586) \\ \ln(GDP_{c,t-1}^{PPP,pc}) -8.714^{***} -7.355^{***} -10.28^{***} \\ (0.833) (1.350) (1.362) \\ BankCrisis_{c,t} -0.547 -0.767 0.579 \\ (0.706) (1.326) (0.868) \\ BankCrisis_{c,t} -0.0375^{**} -0.0538^{***} -0.0343 \\ OutputLoss_{c,t} (0.0153) (0.0208) (0.0224) \\ BankCrisis_{c,t} \times -2.263^{**} 0.651 -7.259^{*} \\ GFC_{c,t} (0.892) (1.210) (3.778) \\ BankCrisis_{c,t} \times 0.916 1.228 1.391 \\ TPI_{c,t-1}^{2010} (0.632) (0.845) (1.132) \\ BankCrisis_{c,t} \times 1.878^{***} 2.794^{***} 1.927^{***} \\ CTPI_{c,t-1}^{2010} (0.325) (0.694) (0.552) \\ \hline N 33,200 16,661 16,539 33,244 16,677 16,561 \\ \hline$	$EFD_i \times Output$	-0.0337	-0.0778	-0.0138	-0.0299	-0.0845*	-0.0105
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$Loss_{c,t}$	(0.0387)	(0.0485)	(0.0558)	(0.0355)	(0.0442)	(0.0603)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$EFD_i \times GFC_{c,t}$	6.284***	2.954	-1.320	9.162***	5.209*	4.230
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(2.149)	(3.192)	(8.257)	(2.002)	(3.110)	(7.244)
$EFD_{i} \times CTPI_{c,t-1}^{2010} -2.597^{**} -1.442 -1.896 -2.229^{**} -0.222 -1.810 \\ (1.188) (1.471) (1.364) (1.038) (1.091) (1.375) \\ ShareManVA_{i,c,t-1} -0.210^{***} -0.187^{****} -0.214^{****} -0.190^{****} -0.190^{****} -0.189^{****} \\ (0.0442) (0.0491) (0.0621) (0.0433) (0.0648) (0.0586) \\ \ln(GDP_{c,t-1}^{PPP,pc}) -8.714^{****} -7.355^{****} -10.28^{****} \\ (0.833) (1.350) (1.362) \\ BankCrisis_{c,t} -0.547 -0.767 0.579 \\ (0.706) (1.326) (0.868) \\ BankCrisis_{c,t} \times -0.0375^{***} -0.0538^{****} -0.0343 \\ OutputLoss_{c,t} (0.0153) (0.0208) (0.0224) \\ BankCrisis_{c,t} \times -2.263^{***} 0.651 -7.259^{**} \\ GFC_{c,t} (0.892) (1.210) (3.778) \\ BankCrisis_{c,t} \times 0.916 1.228 1.391 \\ TPI_{c,t}^{2010} (0.632) (0.845) (1.132) \\ BankCrisis_{c,t} \times 1.878^{***} 2.794^{***} 1.927^{***} \\ CTPI_{c,t-1}^{2010} (0.325) (0.694) (0.552) \\ \hline N 33,200 16,661 16,539 33,244 16,677 16,561 \\ \hline$	$EFD_i \times TPI_{c,t}$	-1.774	-3.508*	-0.0610	-2.358*	-3.997***	-0.994
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\times 1(t \ge 2010)$	(1.328)	(2.017)	(2.194)	(1.371)	(1.244)	(2.426)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$EFD_i \times CTPI_{c,t-1}^{2010}$	-2.597**	-1.442	-1.896	-2.229**	-0.222	-1.810
$\begin{array}{c} (0.0442) (0.0491) (0.0621) (0.0433) (0.0648) \qquad (0.0586) \\ \ln(GDP_{c,t-1}^{PPP,pc}) & -8.714^{***} -7.355^{***} -10.28^{***} \\ (0.833) (1.350) (1.362) \\ BankCrisis_{c,t} & -0.547 -0.767 0.579 \\ (0.706) (1.326) (0.868) \\ BankCrisis_{c,t} \times & -0.0375^{**} -0.0538^{***} -0.0343 \\ OutputLoss_{c,t} (0.0153) (0.0208) (0.0224) \\ BankCrisis_{c,t} \times & -2.263^{**} 0.651 -7.259^{*} \\ GFC_{c,t} (0.892) (1.210) (3.778) \\ BankCrisis_{c,t} \times 0.916 1.228 1.391 \\ TPI_{c,t}^{2010} (0.632) (0.845) (1.132) \\ BankCrisis_{c,t} \times 1.878^{***} 2.794^{***} 1.927^{***} \\ CTPI_{c,t-1}^{2010} (0.325) (0.694) (0.552) \\ \hline N 33,200 16,661 16,539 33,244 16,677 16,561 \\ \hline \end{array}$,	(1.188)	(1.471)	(1.364)	(1.038)	(1.091)	(1.375)
$\begin{array}{c} (0.0442) (0.0491) (0.0621) (0.0433) (0.0648) \qquad (0.0586) \\ \ln(GDP_{c,t-1}^{PPP,pc}) & -8.714^{***} -7.355^{***} -10.28^{***} \\ (0.833) (1.350) (1.362) \\ BankCrisis_{c,t} & -0.547 -0.767 0.579 \\ (0.706) (1.326) (0.868) \\ BankCrisis_{c,t} \times & -0.0375^{**} -0.0538^{***} -0.0343 \\ OutputLoss_{c,t} (0.0153) (0.0208) (0.0224) \\ BankCrisis_{c,t} \times & -2.263^{**} 0.651 -7.259^{*} \\ GFC_{c,t} (0.892) (1.210) (3.778) \\ BankCrisis_{c,t} \times 0.916 1.228 1.391 \\ TPI_{c,t}^{2010} (0.632) (0.845) (1.132) \\ BankCrisis_{c,t} \times 1.878^{***} 2.794^{***} 1.927^{***} \\ CTPI_{c,t-1}^{2010} (0.325) (0.694) (0.552) \\ \hline N 33,200 16,661 16,539 33,244 16,677 16,561 \\ \hline \end{array}$	$ShareManVA_{i,c,t-1}$	-0.210***	-0.187***	-0.214***	-0.190***	-0.190***	-0.189***
$BankCrisis_{c,t} = \begin{pmatrix} (0.833) & (1.350) & (1.362) \\ -0.547 & -0.767 & 0.579 \\ (0.706) & (1.326) & (0.868) \end{pmatrix}$ $BankCrisis_{c,t} \times -0.0375^{**} -0.0538^{***} -0.0343$ $OutputLoss_{c,t} = (0.0153) & (0.0208) & (0.0224)$ $BankCrisis_{c,t} \times -2.263^{**} & 0.651 & -7.259^{*}$ $GFC_{c,t} = (0.892) & (1.210) & (3.778)$ $BankCrisis_{c,t} \times 0.916 & 1.228 & 1.391$ $TPI_{c,t}^{2010} = (0.632) & (0.845) & (1.132)$ $BankCrisis_{c,t} \times 1.878^{***} & 2.794^{***} & 1.927^{***}$ $CTPI_{c,t-1}^{2010} = (0.325) & (0.694) & (0.552)$ $N = 33,200 & 16,661 & 16,539 & 33,244 & 16,677 & 16,561$		(0.0442)	(0.0491)	(0.0621)	(0.0433)	(0.0648)	(0.0586)
$BankCrisis_{c,t} = \begin{pmatrix} (0.833) & (1.350) & (1.362) \\ -0.547 & -0.767 & 0.579 \\ (0.706) & (1.326) & (0.868) \end{pmatrix}$ $BankCrisis_{c,t} \times -0.0375^{**} -0.0538^{***} -0.0343$ $OutputLoss_{c,t} = (0.0153) & (0.0208) & (0.0224)$ $BankCrisis_{c,t} \times -2.263^{**} & 0.651 & -7.259^{*}$ $GFC_{c,t} = (0.892) & (1.210) & (3.778)$ $BankCrisis_{c,t} \times 0.916 & 1.228 & 1.391$ $TPI_{c,t}^{2010} = (0.632) & (0.845) & (1.132)$ $BankCrisis_{c,t} \times 1.878^{***} & 2.794^{***} & 1.927^{***}$ $CTPI_{c,t-1}^{2010} = (0.325) & (0.694) & (0.552)$ $N = 33,200 & 16,661 & 16,539 & 33,244 & 16,677 & 16,561$	$\ln(GDP_{c,t-1}^{PPP,pc})$	-8.714***	-7.355***	-10.28***			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	- 	(0.833)	(1.350)	(1.362)			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$BankCrisis_{c,t}$	-0.547	-0.767	0.579			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.706)	(1.326)	(0.868)			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$BankCrisis_{c,t} \times$	-0.0375**	-0.0538***	-0.0343			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$OutputLoss_{c,t}$	(0.0153)	(0.0208)	(0.0224)			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$BankCrisis_{c,t} \times$	-2.263**	0.651	-7.259*			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$GFC_{c,t}$	(0.892)	(1.210)	(3.778)			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$BankCrisis_{c,t} \times$	0.916	1.228	1.391			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$TPI_{c,t}^{2010}$	(0.632)	(0.845)	` /			
N 33,200 16,661 16,539 33,244 16,677 16,561	$BankCrisis_{c,t} \times$	1.878***	2.794***	1.927***			
N 33,200 16,661 16,539 33,244 16,677 16,561	$CTPI_{c,t-1}^{2010}$	(0.325)	(0.694)	(0.552)			
	N	33,200	16,661	16,539	33,244	16,677	16,561
R^2 (overall) 0.168 0.276 0.124 0.358 0.456 0.333	R^2 (overall)	0.168	0.276	0.124	0.358	0.456	0.333

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

I now summarize in Table 12 the total effect on manufacturing growth from the regressions with macroprudential policies and output losses of banking crises. The results show the average impact on manufacturing growth of a banking crises across three different periods: before the GFC (until 2006), during the GFC (between 2007 and 2009) and post GFC (between 2010 and 2017). In the case of the last period I also show a counterfactual scenario in which no macroprudential policies

had been implemented (that is, considering $TPI_{c,t} = 0$ and $CTPI_{c,t-1}^{2010} = 0$). The triple fixed effects estimates consider only the impact on manufacturing growth through the external finance dependence channel, while the results based on macroeconomic controls consider the total effect on manufacturing growth.

Table 12: Effect on total manufacturing growth (in %, yearly) of banking crises across different periods and policies

criscs deress different periods and penetes								
Coefficient average	е Т	riple F	E (i,c+i,t	(+c,t)	FE (i,c+t) plus macro controls			
across countries	pre-GF0	C GFC 1	post-GFC	C post-GFC*	pre-GFO	C GFC p	ost-GFC	C post-GFC*
		With a	verage w	eights $\varpi_{i,c,t}$	and outp	out loss	for perio	ods t
All countries	-2.5	0.2	-4.4	-2.2	-4.6	-4.2	-1.5	-5.8
AEs	-1.4	0.4	-2.5	-1.6	-4.4	-2.3	-4.1	-6.3
EMDEs	-3.0	-3.7	-5.0	-2.6	-3.9	-6.8	3.2	-5.1

Note 1: post-GFC* is the counterfactual scenario in which no macroprudential policies had been

implemented since 2010. Note 2: FE denotes the set of fixed effects, with i,c+t being industry-country and time, while i,c+i,t+c,t denotes industry-country, industry-time and country-time.

The results show that an absence of macroprudential policies would have improved the growth of externally dependent industries during banking crises. Considering just the external finance dependence channel, banking crises in the post-GFC period would have caused a loss of just 2.2% in manufacturing growth (instead of the 4.4% loss with the macroprudential policies). This loss in growth through the external finance dependence channel happens across all country groups (all countries, AEs and EMDEs). However, when considering the effect on manufacturing through both channels (the direct effect on manufacturing and the external finance dependence channel), the results show that macroprudential policies were highly beneficial for growth during banking crises. In the scenario with no macroprudential policies there is a 5.8% loss in manufacturing growth during banking crises in the post GFC period, which is much higher than the 1.5% growth loss observed with the macroprudential policies implemented. This beneficial effect of macroprudential policies during banking crises in the post-GFC period happens across all country groups. The results show that macroprudential policies increased manufacturing growth during banking crises by 4.3%, 2.2% and 8.3% across the all countries, AEs and EMDEs samples.

7 Conclusions and policy implications

This work studies the effect of banking crises on the real industrial growth, with a particular focus on external finance dependent industries. Using data for 23 industries across 102 countries between 1980 and 2019, I find that banking crises have a negative growth effect on industries through two channels, a direct effect on all manufacturing activity and an effect through industries' external finance dependence. The direct effect of banking crises on manufacturing activity is strong, especially in advanced economies (AEs). The direct effect of a banking crisis on all industries implies a reduction in growth of 2.4% for all countries and 2.5% for AEs. This direct effect of banking crises was even stronger during the Great Financial Crisis (GFC), especially in emerging markets and developing economies (EMDEs). Banking crises during the GFC period implied a further direct loss in manufacturing growth of 2.7%, 2.8% and 9.6% for all countries, AEs and EMDEs, respectively.

The external finance dependence channel is especially large for EMDEs. For the advanced economies the effect is small and not statistically significant. In the model with country-industry, country-year and industry-year fixed effects, fully externally dependent industries during a banking crisis suffer a growth reduction of 3.2%, 1.4% and 7.2% in the samples for all countries, AEs and EMDEs, respectively. The estimates of this effect on manufacturing growth are significantly bigger than in previous studies which ignored the GFC period.

Estimating the effects of other types of financial crises, I find that banking and currency crises have a much stronger effect on externally dependent industries than sovereign debt crises. In the model with country-industry, country-year and industry-year fixed effects, fully externally dependent industries suffer a growth reduction of 3.2%, 4.9% and 1.9% during banking, currency and sovereign debt crises, respectively, in the sample of all countries. Financial crises in EMDEs have an even stronger effect on externally dependent industries. Fully dependent industries in EMDEs reduce their growth by 7.2%, 6.1% and 2.3% during banking, currency and sovereign debt crises. Externally dependent industries in AEs do not experience a statistically significant effect from any type of financial crisis.

I then estimate the effect on total manufacturing growth from each type of financial crises, obtained as the sum of both the effect on all industries and the financial dependence channel. This effect is weighted by the value added of each industry in the national economy. The results show

that currency crises have the strongest effect on manufacturing growth across any country group. For the sample of all countries, manufacturing growth falls by 2.7%, 6% and 1% during banking, currency and sovereign debt crises. EMDEs are the most affected countries by currency crises, while banking and sovereign debt crises impact AEs more strongly. During banking, currency and sovereign debt crises, there is a fall in manufacturing growth of 2.5%, 4.1% and 3.4% for AEs and 2.1%, 6.1% and 0.3% for EMDEs.

Finally, I study the effects of macroprudential policies during banking crises, focusing on the post GFC period when most countries quickly adopted these mechanisms. The results show that macroprudential policies increased manufacturing growth during banking crises in 4.3%, 2.2% and 8.3% for all countries, AEs and EMDEs.

Future research could investigate which other characteristics of the industries, besides external finance dependence, explain vulnerabilities during financial crises in EMDEs. Furthermore, there is a need to measure output effects on other sectors of the economy besides manufacturing.

References

- [1] Akinci, O. and J. Olmstead-Rumsey (2018), "How Effective are Macroprudential Policies? An Empirical Investigation," *Journal of Financial Intermediation*, 33, 33–57.
- [2] Alam, Z., A. Alter, J. Eiseman, G. Gelos, H. Kang, M. Narita, E. Nier and N. Wang (2019), "Digging Deeper Evidence on the Effects of Macroprudential Policies from a New Database," IMF WP/19/66.
- [3] Biljanovska, N., S. Chen, G. Gelos, D. Igan, M. Peria, E. Nier and F. Valencia (2023), "Macroprudential Policy Effects: Evidence and Open Questions," IMF DP 2023/002.
- [4] Bordo, M. and C. Meissner (2016), "Fiscal and Financial Crises," in: J. B. Taylor & Harald Uhlig (ed.), *Handbook of Macroeconomics*, edition 1, volume 2, chapter 7, 355-412.
- [5] Braun, M. and B. Larrain (2005), "Finance and the business cycle: international, inter-industry evidence," *Journal of Finance*, 60(3), 1097–1128.

- [6] Bussière, M., J. Cao, J. de Haan, R. Hills, S. Lloyd, B. Meunier, J. Pedrono, D. Reinhardt, S. Sinha, R. Sowerbutts and K. Styrin (2021), "The interaction between macroprudential policy and monetary policy: Overview," Review of International Economics, 29(1), 1-19.
- [7] Correia, S. (2017), "Linear Models with High-Dimensional Fixed Effects: An Efficient and Feasible Estimator," working Paper.
- [8] Cowan, K. and C. Raddatz (2013), "Sudden stops and financial frictions: Evidence from industry-level data," *Journal of International Money and Finance*, 32, 99-128.
- [9] Dell'Ariccia, G., E. Detragiache and R. Rajan (2008), "The real effect of banking crises," Journal of Financial Intermediation, 17(1), 89-112.
- [10] Frankel, J. (2010), "Monetary Policy in Emerging Markets," in: Benjamin M. Friedman & Michael Woodford (ed.), Handbook of Monetary Economics, edition 1, volume 3, chapter 25, 1439-1520.
- [11] Galati, G. and R. Moessner (2018), "What Do We Know About the Effects of Macroprudential Policy?" *Economica*, 85(340), 735–70.
- [12] Hoggarth, G., R. Reis, Ricardo and V. Saporta (2002), "Costs of banking system instability: Some empirical evidence," *Journal of Banking & Finance*, 26(5), 825-855.
- [13] Igan, D., A. Kutan and A. Mirzaei (2020), "The real effects of capital inflows in emerging markets," *Journal of Banking & Finance*, 119, 105933.
- [14] Igan, D., A. Lauwers and D. Puy (2022), "Capital flows and institutions," BIS Working Papers 994.
- [15] Joyce, J. and M. Nabar (2009), "Sudden stops, banking crises and investment collapses in emerging markets," *Journal of Development Economics*, 90(2), 314-322.
- [16] Kaminsky, G. and C. Reinhart (1999), "The Twin Crises: The Causes of Banking and Balance-of-Payments Problems," American Economic Review, 89(3), 473-500.
- [17] Kroszner, R., L. Laeven and D. Klingebiel (2007), "Banking crises, financial dependence, and growth," *Journal of Financial Economics*, 84(1), 187-228.

- [18] Laeven, L. and F. Valencia (2020), "Systemic Banking Crises Database II," IMF Economic Review, 68(2), 307-361.
- [19] Levchenko, A., R. Rancière and M. Thoenig (2009), "Growth and risk at the industry level: The real effects of financial liberalization," *Journal of Development Economics*, 89(2), 210-222.
- [20] Levine, R. (2005), "Finance and growth: Theory, evidence, and mechanisms," in The Handbook of Economic Growth, ed. P. Aghion and S. Durlauf, Amsterdam: North-Holland.
- [21] Linde, J., F. Smets and R. Wouters (2016), "Challenges for Central Banks' Macro Models," in: J. B. Taylor & Harald Uhlig (ed.), *Handbook of Macroeconomics*, edition 1, volume 2, chapter 28, 2185-2262.
- [22] Machado, J. and J. Santos Silva (2019), "Quantiles via moments," *Journal of Econometrics*, 213(1), 145-173.
- [23] Madeira, C. and J. Madeira (2019), "The effect of FOMC votes on financial markets", Review of Economics and Statistics, 101(5), 921-932.
- [24] Madeira, C., J. Madeira and P. Monteiro (2023), "The origins of monetary policy disagreement: the role of supply and demand shocks", *Review of Economics and Statistics*, forthcoming.
- [25] Madeira, C. (2024), "The impact of macroprudential policies on industrial growth," BIS Working Papers.
- [26] Mendoza, E. and K. Smith (2006), "Quantitative implications of a debt-deflation theory of Sudden Stops and asset prices," *Journal of International Economics*, 70(1), 82-114.
- [27] Nakatani, R. (2020), "Macroprudential policy and the probability of a banking crisis," Journal of Policy Modeling, 42(6), 1169-1186.
- [28] Nguyen, T., V. Castro and J. Wood (2022), "A new comprehensive database of financial crises: Identification, frequency, and duration," *Economic Modelling*, 108, 105770.
- [29] Raddatz, C. (2006), "Liquidity needs and vulnerability to financial underdevelopment," Journal of Financial Economics, 80(3), 677–722.

- [30] Raddatz, C. (2010), "Credit Chains and Sectoral Comovement: Does the Use of Trade Credit Amplify Sectoral Shocks?," *Review of Economics and Statistics*, 92(4), 985-1003.
- [31] Rajan, R. and L. Zingales (1998), "Financial Dependence and Growth," *American Economic Review*, 88(3), 559–586.
- [32] Rodrik, D. (2021), "Why Does Globalization Fuel Populism? Economics, Culture, and the Rise of Right-Wing Populism," *Annual Review of Economics*, 13(1), 133-170.
- [33] Schularick, M. and A. Taylor (2012), "Credit Booms Gone Bust: Monetary Policy, Leverage Cycles, and Financial Crises, 1870-2008," *American Economic Review*, 102 (2), 1029-1061.

The impact of financial crises on industrial growth: lessons from the last 40 years

Appendix A: Adding controls for real GDP growth, inflation, monetary policy rates and its lags

This appendix shows results that add the real GDP growth, inflation and monetary policy rates and its lags as additional controls. It also adds the duration, fiscal costs and peak liquidity delivered to the financial system during the crises as additional controls.

Furthermore, the emerging markets and developing economies (EMDEs) country category is separated into 2 groups with a similar number of countries: emerging markets (EMs) and low income countries (LICs). Therefore the study comprises 102 countries: 35 advanced economies, 34 emerging markets and 33 low-income countries. Low income countries are defined as countries that have an average GDP per capita for the period 2011-2021 that is lower than 11,000 USD (in constant 2017 prices). Some exceptions are made for large economies that are traditionally classified as emerging markets instead.

The results show that the current year's GDP growth effect on industrial growth is positive and statistically significant, whatever the country sample. The effect of GDP growth on industrial growth is slightly larger than one in AEs and EMs, while being a bit lower than one for LICs. This shows that the manufacturing sector has a higher business cycle component than the average firm of the economy in AEs and EMs, while being less cyclical in LICs.

Table A.1: Industries and countries available in the joint industrial and macroprudential policy dataset

Industries (ISIC 2-digit revision 3) with External Finance Dependence (EFD_i) in parentheses: 15 Food and beverages (0.112), 16 Tobacco products (-0.451), 17 Textiles (0.277), 18 Wearing apparel, fur (0.029), 19 Leather, leather products and footwear (-0.113), 20 Wood products (excl. furniture) (0.283), 21 Paper and paper products (0.161), 22 Printing and publishing (0.203), 23 Coke, refined petroleum products, nuclear fuel (0.170), 24 Chemicals and chemical products (0.458), 25 Rubber and plastics products (0.634), 26 Non-metallic mineral products (0.193), 27 Basic metals (0.040), 28 Fabricated metal products (0.213), 29 Machinery and equipment n.e.c. (0.633), 30 Office, accounting and computing machinery (0.948), 31 Electrical machinery and apparatus (0.821), 32 Radio, television and communication equipment (0.975), 33 Scientific instruments, medical, precision and optical instruments (0.961), 34 Motor vehicles, trailers, semi-trailers (0.360), 35 Other transport equipment (0.328), 36 Furniture; manufacturing n.e.c. (0.235), 37 Other manufactured products and recycling (0.339).

Countries covered (102). Advanced Economies (35): Australia, Austria, Belgium, Canada, Cyprus, Czechia, Denmark, Estonia, Finland, France, Germany, Greece, Hong Kong, Iceland, Ireland, Israel, Italy, Japan, South Korea, Latvia, Lithuania, Luxembourg, Netherlands, New Zealand, Norway, Portugal, Singapore, Slovakia, Slovenia, Spain, Sweden, Switzerland, Taiwan, UK, USA.

Emerging markets (34): Algeria, Argentina, Barbados, Belarus, Bosnia, Brazil, Bulgaria, Chile, China, Colombia, Costa Rica, Croatia, Dominican Republic, Hungary, Iraq, Kuwait, Lebanon, Macao, Malaysia, North Macedonia, Mauritius, Mexico, Oman, Panama, Peru, Poland, Romania, Russian Federation, Serbia, South Africa, Thailand, Trinidad and Tobago, Turkey, Uruguay.

Low income countries (33): Albania, Armenia, Bangladesh, Burundi, Cameroon, Ecuador, Egypt, Fiji, Honduras, India, Indonesia, Jamaica, Jordan, Kyrgyzstan, Lao, Lesotho, Madagascar, Moldova, Mongolia, Morocco, Myanmar, Nepal, Pakistan, Paraguay, Philippines, Rwanda, Senegal, Sri Lanka, Tunisia, Vietnam, Ukraine, Uzbekistan, Zambia.

Distribution of the share of manufacturing value-added in GDP across countries in 2019 (in %):															
$Share_{i,c,t}$	\min	p10	p25	p50	p75	p90	max	min	$\mathbf{p}10$	p25	$\mathbf{p50}$	$\mathbf{p75}$	$\mathbf{p90}$	max	
Total manufacturing over GDP Largest national industry over GDP															
All countries	0.6	6.3	9.5	12.7	17.6	21.4	47.6	0.3	1.4	2.0	2.8	3.9	5.1	18.0	
\mathbf{AEs}	1.2	5.8	9.6	12.4	18.7	22.6	31.5	0.4	1.2	1.8	2.1	3.2	5.1	14.6	
\mathbf{EMs}	0.6	5.0	10.4	12.7	18.0	23.9	47.6	0.3	1.3	2.1	2.9	3.9	4.6	18.0	
LICs	5.9	7.6	8.7	13.0	15.6	18.5	26.1	1.0	1.9	2.8	3.1	4.4	6.0	14.9	
Individual industries value-added as a share of GDP in 2019 and correlation of the															

Individual industries value-added as a share of GDP in 2019 and correlation of the individual industries' real growth with real GDP growth during 1980-2019 (in %):

$Share_{i,c,t}$	p10	p25	p50	p75	p90	p95	p99	mın p10 p25 p50 p75 p90 p9	5
	Indi	vidu	al ma	anufa	cture	es ove	r GDP	Correlation with GDP growth	
All countries	0.03	0.1	0.3	0.7	1.5	2.4	4.6	95.7 -11.4 15.4 41.9 63.8 77.6 82.	.9
\mathbf{AEs}	0.04	0.1	0.4	0.8	1.5	2.1	3.9	-65.2 10.1 32.4 49.8 67.3 78.7 83.	.3
\mathbf{EMs}	0.03	0.1	0.3	0.7	1.6	2.4	4.6	-85.9 -5.4 21.0 49.2 66.7 79.1 83.	.8
\mathbf{LICs}	0.03	0.1	0.2	0.6	1.4	2.8	4.9	-95.7 -49.2 -8.7 18.2 42.1 67.6 81.	.4

Table A.2: Extra variables used in the appendix and their sources

Variable	Description	Source
$g_{c,t}^{GDP}$	Real GDP growth rate of country c in year t	World Bank
$MPR_{c,t}$	Monetary policy rate of country c in year t (mean yearly value)	IMF & BIS
$inflation_{c,t}$	Consumer Price Index inflation of country c in year t	World Bank
Crisis	Output loss given by the cumulative sum of the difference between	Laeven &
$OutputLoss_{c,t}$	actual and trend real GDP over the period [T, T+3], in % of GDP	Valencia
Crisis	Duration (in years) of the financial crisis.	Laeven &
$Duration_{c,t}$		Valencia
Crisis	Fiscal costs refer to outlays directly related to the restructuring	Laeven &
$FiscalCosts_{c,t}$	of the financial sector, in $\%$ of GDP	Valencia
Crisis	Liquidity is the ratio (in %) of central bank claims on deposit banks	Laeven &
$PeakLiquidity_{c,t}$	and Treasury liquidity support to total deposits and foreign liabilities	Valencia

Table A.3: Effects of banking crises on industrial growth across country groups (all countries, AEs, EMs, LICs)

Controls	Witl	n macroeco	nomic cont	rols	With country-year fixed effects				
	All		Emerging		All		Emerging	Low	
	countries	economies	markets	income	countries	economies	markets	income	
$EFD_i \times$	-3.420***	-0.908	-8.138***	-4.802	-3.562***	-1.153	-7.943***	-5.487*	
$BankCrisis_{c,t}$	(1.006)	(1.258)	(1.702)	(3.234)	(0.963)	(1.224)	(1.620)	(3.042)	
$ShareManVA_{i,c,t-1}$	-0.161***	-0.144***	-0.171***	-0.168**	-0.154***	-0.136***	-0.150***	-0.185**	
	(0.0340)	(0.0431)	(0.0554)	(0.0798)	(0.0331)	(0.0409)	(0.0551)	(0.0781)	
$\ln(GDP_{c,t-1}^{PPP,pc})$	-2.207***	-2.599***	-3.491***	0.573					
	(0.611)	(0.862)	(1.259)	(1.960)					
$inflation_{c,t}$	-0.000755*	-0.0423	-0.000367	-0.0844**					
,	(0.000389)	(0.0677)	(0.000423)	(0.0397)					
$g_{c,t}^{GDP}$	1.197***	1.107***	1.346***	0.888***					
-,-	(0.0479)	(0.0625)	(0.0821)	(0.109)					
$inflation_{c,t-1}$	0.000122	-0.0130	0.000434	0.00933					
	(0.000467)	(0.0106)	(0.000482)	(0.0273)					
$g_{c,t-1}^{GDP}$	-0.0363	-0.203***	0.0138	-0.0198					
-,	(0.0356)	(0.0497)	(0.0632)	(0.0839)					
$BankCrisis_{c,t}$	0.959**	-0.165	2.554***	1.591					
	(0.378)	(0.493)	(0.663)	(1.327)					
N	41,545	20,228	13,286	8,031	41,980	20,472	13,496	8,012	
R^2 (overall)	0.188	0.279	0.209	0.102	0.315	0.369	0.358	0.215	
· · · · · · · · · · · · · · · · · · ·				/\					

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

All regressions include fixed effects by industry-country and year (omitted).

Table A.4: Effects of banking crises on industrial growth, with controls for monetary policy With monetary policy in levels: With monetary policy increase: Controls $MonPol_{c,t} = \Delta MPR_{c,t}$ $MonPol_{c,t} = MPR_{c,t}$ Advanced Emerging Advanced Emerging Low All Low All countries economies markets income countries economies markets income $\overline{EFD_i} \times$ -9.638*** -10.27*** -10.18*** -9.414** -2.553** 0.360 -2.695** 0.238 $BankCrisis_{c.t}$ (2.136)(1.129)(1.266)(3.817)(1.162)(1.306)(2.178)(3.762) $ShareManVA_{i,c,t-1}$ -0.260*** -0.214*** -0.248** -0.355** -0.283*** -0.240*** -0.231* -0.420*** (0.119)(0.0513)(0.0555)(0.114)(0.143)(0.0546)(0.0594)(0.157) $\ln(GDP_{c,t-1}^{PPP,pc})$ -5.858*** -8.002*** -6.476*** -8.646*** -4.525*** -5.214-5.111*** -5.122(0.919)(1.611)(1.342)(4.625)(0.922)(1.760)(1.499)(4.287) $inflation_{c,t}$ 0.00737*0.345*** 0.1900.0336**0.369*** 0.0372*** 0.001440.184(0.00413)(0.00585)(0.201)(0.0147)(0.206)(0.126)(0.127)(0.0132)1.155*** 0.926*** 1.067*** 1.301*** 1.175*** 0.918*** 1.297*** 1.069*** (0.0597)(0.0785)(0.0880)(0.208)(0.0603)(0.0806)(0.0830)(0.212) $inflation_{c,t-1}$ -0.0378*** -0.353*** -0.0301*** -0.408*** -0.0283*** -0.0286-0.0279** -0.102(0.0108)(0.00485)(0.110)(0.00611)(0.189)(0.0119)(0.114)(0.170) $g_{c,t-1}^{GDP}$ -0.216*** -0.206*** -0.271*** -0.322** -0.194*** -0.202*** -0.212*** -0.310** (0.0459)(0.0769)(0.0678)(0.0462)(0.0689)(0.149)(0.140)(0.0772) $BankCrisis_{c,t}$ -0.371-0.906* 0.6465.387**-0.236-1.055* 1.253 5.756**(0.430)(0.529)(0.754)(2.476)(0.442)(0.546)(0.774)(2.425) $MonPol_{c.t}$ 0.001360.1240.002770.207-0.00911** -0.00959*** 0.06780.184(0.00164)(0.114)(0.00221)(0.255)(0.00396)(0.109)(0.00357)(0.221) $MonPol_{c,t-1}$ 0.0127*** -0.477*** 0.0105*** -0.372*** -0.000558 -0.364-0.000378 -0.107(0.00220)(0.000270)(0.00175)(0.124)(0.249)(0.0984)(0.000431)(0.151) $MonPol_{c,t-2}$ 0.0003650.330*** 0.000421 0.1690.000360-0.207** 0.000610*** 0.0396(0.000386)(0.0965)(0.000430)(0.117)(0.000235)(0.0817)(0.000227)(0.102) $\overline{\mathrm{N}}$ 28,012 15,765 15,238 8,507 3,740 26,906 8,135 3,533 R^2 (overall) 0.2380.2900.2640.1570.2390.2980.2660.146

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

Table A.5: Effects of banking crises on industrial growth, with controls for monetary, fiscal and liquidity policies

Controls Without accounting for monetary policy With monetary policy increase:

Controls	***************************************	, cooding	ioi inonota	ij ponej	• •	MonPol	$c_{c,t} = \Delta MPI$	R _{at}
	All	Advanced	Emerging	Low	All		Emerging	Low
		economies	0 0	income		economies	\max_{t}	income
$EFD_i \times$	-6.620***	-0.915	-11.99**	-8.787	-6.165***	0.201	-21.64***	-7.188
$BankCrisis_{c.t}$	(1.920)	(1.719)	(5.625)	(16.12)	(2.016)	(1.794)	(5.398)	(14.43)
$EFD_i \times Crisis$	-0.0358	-0.157***	-0.0272	0.161	-0.0568	-0.202***	$0.119^{'}$	-0.0751
$OutputLoss_{c,t}$	(0.0395)	(0.0564)	(0.0662)	(0.0977)	(0.0452)	(0.0556)	(0.0799)	(0.220)
$EFD_i \times Crisis$	1.778***	1.761***	1.262	-0.416	2.005***	1.722***	2.395*	
$Duration_{c,t}$	(0.611)	(0.539)	(1.417)	(5.423)	(0.622)	(0.517)	(1.392)	
$EFD_i \times Crisis$	0.00838	0.380**	0.0882	-0.356	0.123	0.426***	0.0904	
$FiscalCosts_{c,t}$	(0.0983)	(0.148)	(0.209)	(0.218)	(0.124)	(0.164)	(0.232)	
$EFD_i \times Crisis$	-0.0558	-0.211***	-0.0433	0.282**	-0.142***	-0.183***	-0.00797	
$PeakLiquidity_{c,t}$	(0.0341)	(0.0518)	(0.0518)	(0.119)	(0.0337)	(0.0542)	(0.0504)	
$ShareManVA_{i,c,t-1}$	-0.134***	-0.130***	-0.111	-0.181**	-0.286***	-0.237***	-0.230*	-0.432**
222	(0.0448)	(0.0462)	(0.0795)	(0.0866)	(0.0589)	(0.0652)	(0.125)	(0.169)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-2.633***	-2.332**	-4.260***	-1.327	-6.308***	-9.558***	-4.852***	-8.360
,	(0.669)	(0.946)	(1.476)	(2.081)	(1.042)	(2.048)	(1.808)	(5.368)
$inflation_{c,t}$	-0.000684*	-0.0578	-0.000202	-0.0850**	0.0274*	0.379***	0.0290**	0.276
	(0.000387)	(0.0551)	(0.000420)	(0.0402)	(0.0153)	(0.126)	(0.0140)	(0.246)
$g_{c,t}^{GDP}$	1.206***	1.107***	1.374***	0.866***	1.181***	0.901***	1.326***	1.014***
	(0.0485)	(0.0653)	(0.0805)	(0.114)	(0.0616)	(0.0827)	(0.0846)	(0.222)
$inflation_{c,t-1}$	8.34e-05	-0.0149	0.000433	0.0131	-0.0228*	-0.446***	-0.0225**	-0.0567
	(0.000474)	(0.0105)	(0.000485)	(0.0266)	(0.0123)	(0.117)	(0.0112)	(0.208)
$g_{c,t-1}^{GDP}$	-0.0430	-0.225***	0.00682	-0.0202	-0.216***	-0.264***	-0.200***	-0.322**
	(0.0371)	(0.0515)	(0.0650)	(0.0868)	(0.0480)	(0.0784)	(0.0739)	(0.161)
$BankCrisis_{c,t}$	0.753*	-0.437	2.596***	1.013	-0.361	-0.894	1.622**	2.920
	(0.412)	(0.500)	(0.750)	(1.654)	(0.458)	(0.555)	(0.813)	(4.434)
$MonPol_{c,t}$					-0.00729*	0.156	-0.00763**	0.114
					(0.00408)	(0.104)	(0.00372)	(0.316)
$MonPol_{c,t-1}$					-0.000298	-0.350***	-0.000571	-0.243
					(0.000307)	(0.0866)	(0.000438)	(0.210)
$MonPol_{c,t-2}$					0.000276		0.000503**	0.0885
					(0.000245)	(0.0763)	(0.000236)	(0.127)
N	37,988	18,758	12,107	7,123	24,147	13,969	7,259	2,919
R^2 (overall)	0.200	0.308	0.220	0.107	0.262	0.333	0.280	0.163

Robust standard errors in (). Clusters by industry-country.

^{***,**,*} denote 1%, 5%, 10% statistical significance.

Table A.6: Effects of banking crises on industrial growth, with interactions for different periods (Great Moderation, 1988-2006; Great Financial Crisis, 2007-2009)

Controls Without accounting for monetary policy With monetary policy increase: $MonPol_{c,t} = \Delta MPR_{c,t}$ All Advanced Emerging Low All Advanced Emerging Low countries markets economies markets income countries economies income $EFD_i \times BankCrisis_{c.t}$ -12.71** -0.4690.306-9.621* -1.654-0.3560.397-6.975(1.650)(1.870)(5.841)(4.724)(1.788)(1.922)(5.953)(5.572) $EFD_i \times BankCrisis_{c.t.}$ -3.062* -0.1922.603 -4.028-3.2763.656-4.5514.028 $\times GreatModeration_t$ (1.831)(2.467)(5.896)(5.181)(2.172)(2.710)(6.091)(7.059) $EFD_i \times BankCrisis_{c.t.}$ -5.808*** -2.899-6.545-44.57-3.750** -1.589-2.678 $\times GFC_t$ (1.683)(1.801)(5.502)(43.87)(1.702)(1.824)(5.548) $ShareManVA_{i,c,t-1}$ -0.162*** -0.144*** -0.171*** -0.170** -0.286*** -0.238*** -0.229** -0.421*** (0.0340)(0.0430)(0.0551)(0.0790)(0.0549)(0.0590)(0.116)(0.157) $\ln(GDP_{c,t-1}^{PPP,pc})$ -2.176*** -2.556*** -3.451*** -8.382*** -4.386*** 0.377-5.072*** -4.931 (0.611)(0.862)(1.258)(1.996)(0.922)(1.759)(4.320)(1.511)0.357*** $inflation_{c,t}$ -0.000744* -0.0442-0.000373 -0.0850** 0.0337**0.0374***0.198(0.000390)(0.0661)(0.000422)(0.0397)(0.0147)(0.129)(0.0132)(0.203) $g_{c,t}^{GDP}$ 1.114*** 0.917*** 1.198*** 1.346*** 0.883***1.176*** 1.296*** 1.067*** (0.0478)(0.0625)(0.0821)(0.108)(0.0603)(0.0806)(0.0833)(0.212)-0.424*** -0.0286*** $inflation_{c,t-1}$ 0.000125-0.01280.0004490.00855-0.0278** -0.0900(0.000466)(0.0106)(0.000480)(0.0276)(0.0119)(0.0107)(0.173)(0.115)-0.199*** -0.0121-0.192*** -0.199** -0.209*** -0.314** -0.03340.0171(0.0356)(0.0500)(0.0631)(0.0837)(0.0463)(0.0775)(0.0691)(0.149)1.380*0.929**2.553*** $BankCrisis_{c,t}$ -0.2642.014-0.255-1.272** 5.288** (0.382)(0.511)(0.660)(1.350)(0.445)(0.746)(2.552)(0.556) $MonPol_{c,t}$ -0.00914** -0.00961*** 0.1810.0745(0.221)(0.00396)(0.110)(0.00356) $MonPol_{c,t-1}$ -0.359*** -0.000426 -0.000482-0.120(0.000271)(0.0986)(0.000450)(0.154) $MonPol_{c,t-2}$ -0.197** 0.000618*** 0.0003580.0395(0.000235)(0.0819)(0.000228)(0.102)20,228 13,286 8,031 41,545 26,906 15,238 8,135 3,533 R^2 (overall) 0.2790.2090.2400.2980.2660.1880.1030.146

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

Table A.7: Effects of different types of financial crises (banking, currency, sovereign debt, any financial crisis), across the entire country sample

Controls	Wit	h macroeco	nomics con	trols	W	ith countr	y-year fixed	effects
$EFD_i \times Banking$				-2.869***				-2.993***
$Crisis_{c,t}$				(1.036)				(0.990)
$EFD_i \times Currency$	-4.894***			-3.835**	-5.048***			-4.034**
$Crisis_{c,t}$	(1.745)			(1.812)	(1.794)			(1.871)
$EFD_i \times Sovereign$		-2.471**		-1.468		-2.135*		-1.076
$DebtCrisis_{c,t}$		(1.249)		(1.271)		(1.126)		(1.155)
$EFD_i \times Any$			-2.739***				-2.832***	
$Crisis_{c,t}$			(0.851)				(0.821)	
$ShareManVA_{i,c,t-1}$	-0.160***	-0.160***	-0.162***	-0.161***	-0.152***	-0.153***	-0.154***	-0.154***
	(0.0339)	(0.0338)	(0.0340)	(0.0341)	(0.0330)	(0.0329)	(0.0331)	(0.0332)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-2.161***	-2.172***	-2.274***	-2.175***				
-,	(0.607)	(0.612)	(0.616)	(0.614)				
$inflation_{c,t}$	-0.000456	-0.000750*	-0.000744*	-0.000466				
	(0.000393)	(0.000394)	(0.000389)	(0.000398)				
$g_{c,t}^{GDP}$	1.172***	1.200***	1.189***	1.171***				
,	(0.0481)	(0.0473)	(0.0482)	(0.0488)				
$inflation_{c,t-1}$	3.64 e - 05	0.000132	0.000167	1.53 e-05				
	(0.000469)	(0.000464)	(0.000464)	(0.000468)				
$g_{c,t-1}^{GDP}$	-0.0305	-0.0349	-0.0401	-0.0298				
,	(0.0352)	(0.0355)	(0.0357)	(0.0356)				
$BankCrisis_{c,t}$				0.906**				
				(0.386)				
$CurrencyCrisis_{c,t}$	-0.248			-0.588				
	(0.790)			(0.812)				
$DebtCrisis_{c,t}$		0.724		0.581				
		(0.450)		(0.464)				
$AnyCrisis_{c,t}$			0.430					
			(0.359)					
N	$41,\!545$	$41,\!545$	41,545	41,545	41,980	41,980	41,980	41,980
R^2 (overall)	0.188	0.187	0.188	0.188	0.315	0.315	0.315	0.316

Robust standard errors in (). Clusters by industry-country.

***,**,* denote 1%, 5%, 10% statistical significance.

Table A.8: Effects of different types of financial crises across country groups (AEs, EMs, LICs)

Controls With macro controls With country-year fixed effects

Advanced Emerging Low Advanced Emerging Low	
Mavaneed Emerging Low Mavaneed Emerging Low	
economies markets income economies markets income	
$EFD_i \times Banking$ -1.126 -7.574*** -2.541 -1.137 -7.411*** -3.315	
$Crisis_{c,t}$ (1.242) (1.830) (3.345) (1.217) (1.678) (3.234)	
$EFD_i \times Currency$ 5.207 -2.617* -9.236 1.075 -2.377 -9.249	
$Crisis_{c,t}$ (3.639) (1.493) (6.155) (4.709) (1.504) (6.601)	
$EFD_i \times Sovereign$ -4.179 0.187 -1.307 -4.035 0.130 -0.951	
$DebtCrisis_{c,t}$ (5.066) (1.404) (2.173) (4.276) (1.227) (2.169)	
$ShareManVA_{i,c,t-1} - 0.144*** - 0.170*** - 0.170*** - 0.137*** - 0.150*** - 0.186**$	
(0.0432) (0.0554) (0.0798) (0.0410) (0.0552) (0.0783)	
$\ln(GDP_{c,t-1}^{PPP,pc})$ -2.653*** -4.044*** 0.203	
(0.865) (1.292) (1.962)	
$inflation_{c,t}$ -0.0399 -8.10e-05 -0.0727*	
(0.0676) (0.000427) (0.0410)	
$g_{c,t}^{GDP}$ 1.100*** 1.315*** 0.849***	
(0.0621) (0.0847) (0.113)	
$inflation_{c,t-1}$ -0.0124 0.000509 0.00139	
(0.0107) (0.000471) (0.0277)	
$g_{c,t-1}^{GDP}$ -0.205*** 0.0249 -0.0193	
(0.0498) (0.0630) (0.0831)	
$BankCrisis_{c,t}$ -0.0126 2.655*** 1.128	
$(0.493) \qquad (0.691) \qquad (1.392)$	
$CurrencyCrisis_{c,t}$ -3.613** -0.441 0.284	
$(1.723) \qquad (0.730) \qquad (2.552)$	
$DebtCrisis_{c,t}$ 0.414 -1.476** 1.075	
$(1.383) \qquad (0.621) \qquad (0.880)$	
N 20,228 13,286 8,031 20,472 13,496 8,012	
R^2 (overall) 0.279 0.209 0.104 0.369 0.358 0.216	

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

Table A.9: Quantile regressions with country-industry fixed effects for the impact of different types of financial crises (banking, currency, sovereign debt)

<i>J</i> 1	All countries			Advanced	Emerging	Low
				economies	\max	income
Controls	Q25	Q50	Q75	Q50	Q50	Q50
$EFD_i \times Banking$	-3.301*	-2.923	-2.556	-1.240	-7.469*	-2.747
$Crisis_{c,t}$	(1.845)	(2.756)	(4.566)	(1.131)	(3.967)	(6.267)
$EFD_i \times Currency$	-3.347	-3.738	-4.117	5.288	-2.743	-9.478
$Crisis_{c,t}$	(2.873)	(4.290)	(7.109)	(3.817)	(3.146)	(7.508)
$EFD_i \times Sovereign$	-3.070	-1.479	0.0643	-4.055	0.238	-1.327
$DebtCrisis_{c,t}$	(2.339)	(3.493)	(5.787)	(4.408)	(2.955)	(3.809)
$ShareManVA_{i,c,t-1}$	-0.0532	-0.162*	-0.268*	-0.141***	-0.173	-0.173
	(0.0595)	(0.0888)	(0.147)	(0.0292)	(0.118)	(0.115)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-0.524	-3.014**	-5.429***	-2.559***	-3.600*	-2.876
3,0 1	(0.837)	(1.248)	(2.067)	(0.464)	(2.049)	(2.549)
$inflation_{c,t}$	-0.000465	-8.03e-05	0.000293	0.133***	0.000287	-0.0673
	(0.00127)	(0.00190)	(0.00315)	(0.0417)	(0.00125)	(0.0690)
$g_{c,t}^{GDP}$	1.486***	1.357***	1.231***	1.475***	1.426***	0.924***
,	(0.0601)	(0.0896)	(0.148)	(0.0393)	(0.101)	(0.156)
$inflation_{c,t-1}$	-0.000268	-0.000274	-0.000280	-0.0436***	-8.14e-06	-0.00625
	(0.000969)	(0.00145)	(0.00240)	(0.0140)	(0.000960)	(0.0589)
$g_{c,t-1}^{GDP}$	0.00235	-0.0917	-0.183	-0.291***	-0.0320	-0.0236
,	(0.0549)	(0.0820)	(0.136)	(0.0352)	(0.0966)	(0.145)
$BankCrisis_{c,t}$	0.259	0.968	1.655	0.308	2.444*	1.449
	(0.740)	(1.105)	(1.831)	(0.458)	(1.481)	(2.313)
$CurrencyCrisis_{c,t}$	-0.250	0.191	0.618	-1.757	0.267	0.644
	(1.115)	(1.666)	(2.760)	(1.583)	(1.244)	(2.820)
$DebtCrisis_{c,t}$	1.727*	0.744	-0.210	0.285	-0.888	2.076
	(0.888)	(1.326)	(2.197)	(1.273)	(1.224)	(1.501)
$GreatModeration_t$	-1.740***	-1.389**	-1.049	-1.490***	-1.371	-1.069
	(0.383)	(0.572)	(0.948)	(0.192)	(0.972)	(1.538)
GFC_t	-5.222***	-3.438***	-1.709	-3.818***	-3.506***	-1.760
	(0.523)	(0.779)	(1.291)	(0.318)	(1.024)	(1.519)
N	41,545	41,545	41,545	20,228	13,286	8,031

Standard errors in ().

***, **, * denote 1%, 5%, 10% statistical significance.

The impact of financial crises on industrial growth: lessons from the last 40 years

Appendix B: adding the inflation and monetary policy rates as additional controls

This work shows the impact of financial crises across industries and total manufacturing sector. Externally dependent industries experience lower growth during banking crises. This effect is twice as strong in emerging markets and was significantly worse during the Great Financial Crisis. Banking crises' effects on emerging markets' externally dependent industries are twice as strong as currency crises. In low-income countries, currency crises show the strongest effect, but with high statistical uncertainty. Sovereign debt crises have a low statistical significance across all country groups. Externally dependent industries in advanced economies do not suffer statistically significant growth effects during financial crises.

This appendix shows results that add the inflation and monetary policy rates and its lags as additional controls. It also adds the duration, fiscal costs and peak liquidity delivered to the financial system during the crises as additional controls.

Furthermore, the emerging markets and developing economies (EMDEs) country category is separated into 2 groups with a similar number of countries: emerging markets (EMs) and low income countries (LICs).

The results are fairly similar to the main article: i) the inflation and monetary policy rates have small coefficients; ii) LICs represent just one third of the observations for the EMDEs, therefore the group EMs has very similar results to the EMDEs.

Table B.2 shows that for AEs current inflation increases the industrial growth rate, but lagged inflation decreases it. The total effect of inflation (that is, the sum of current and lagged coefficients) is negative for industrial growth in AEs. This result makes sense in the light of New Keynesian models with price rigidities.

Accounting for the severity and duration of the crisis can be important (Laeven and Valencia 2020). The effect of the duration of the banking crisis on the growth of externally dependent industries is positive in the samples of all countries and AEs. This result could be due to financially dependent industries accumulating a buffer for investment as a crisis gets longer and therefore being gradually less impacted.

Furthermore, the policy effects analysis in Table B.3 shows that fiscal expenses are associated with a lower growth of externally dependent industries in the samples of all countries and for LICs. This effect in LICs could be due to the worse governance of its institutions and insolvency risk. Liquidity support to financial institutions during banking crises has a negative effect on the samples of all countries, AEs and EMs, for the models with monetary policy controls. Without monetary policy controls, liquidity support has a negative effect on industrial growth in the all countries and AEs samples. It is worth noting that Dell'Ariccia et al. 2008 also found a negative effect of liquidity support for their sample of 41 countries, although without statistical significance. My analysis of the effects of fiscal expenses during banking crises is new and therefore cannot be compared to results in the previous literature. The negative effects of fiscal policy and liquidity support point towards the need of improving policy governance before financial crises happen.

For the model with monetary policy controls, I find that for AEs current inflation increases the industrial growth rate, but lagged inflation decreases it. The total effect of inflation (that is, the sum of current and lagged coefficients) is negative for industrial growth in AEs. This result makes sense in the light of New Keynesian models with price rigidities. This result differs from the models that do not include monetary policy controls in Table B.3, which fail to take into account that central banks may increase monetary policy and reduce growth during periods of high inflation.

Table B.1: Effects of banking crises on industrial growth across country groups Controls With macroeconomic controls

0 01101 010		111011 111		001101 010
	All	Advanced	Emerging	Low
	countries	economies	\max	income
$EFD_i \times$	-3.377***	-0.852	-8.467***	-4.675
$BankCrisis_{c,t}$	(1.080)	(1.321)	(1.907)	(3.384)
$ShareManVA_{i,c,t-1}$	-0.164***	-0.152***	-0.164***	-0.172**
	(0.0360)	(0.0439)	(0.0603)	(0.0786)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.748***	-7.665***	-8.828***	-1.230
,	(0.596)	(0.820)	(1.301)	(1.944)
$inflation_{c,t}$	-0.00357***	-0.148**	-0.00341***	-0.147***
	(0.000421)	(0.0689)	(0.000434)	(0.0394)
$inflation_{c,t-1}$	0.000287	-0.00540	0.000993**	0.0317
	(0.000471)	(0.0116)	(0.000478)	(0.0280)
$BankCrisis_{c,t}$	-2.333***	-2.264***	-1.623**	0.249
	(0.397)	(0.512)	(0.722)	(1.336)
N	41,545	20,228	13,286	8,031
R^2 (overall)	0.143	0.244	0.142	0.089
		. ()		

Robust standard errors in (). Clusters by industry-country.

***,**,* denote 1%, 5%, 10% statistical significance.

Table B.2: Effects of banking crises on industrial growth, with controls for monetary policy With monetary policy increase: Controls With monetary policy in levels: $MonPol_{c,t} = MPR_{c,t}$ $MonPol_{c,t} = \Delta MPR_{c,t}$ Advanced Emerging Advanced Emerging Low All Low All countries economies markets income countries economies markets income $EFD_i \times$ -2.284* -9.732*** -10.26*** -2.391* -10.27*** -9.262** 0.4770.370 $BankCrisis_{c,t}$ (1.186)(1.324)(2.189)(3.839)(1.226)(1.356)(2.248)(3.802) $ShareManVA_{i,c,t-1}$ -0.262*** -0.208*** -0.239** -0.352** -0.286*** -0.232*** -0.222* -0.423*** (0.0532)(0.0577)(0.119)(0.147)(0.0573)(0.0619)(0.126)(0.163) $\ln(GDP_{c,t-1}^{PPP,pc})$ -11.46*** -13.34*** -15.56*** -13.44*** -10.31*** -14.18*** -13.22*** -14.54*** (0.958)(5.022)(0.979)(4.698)(1.653)(1.465)(1.851)(1.582)0.515*** $inflation_{c,t}$ 0.480***-0.0143** -0.00500 0.127-0.004470.002450.114(0.00611)(0.00636)(0.0182)(0.120)(0.200)(0.121)(0.0161)(0.201) $inflation_{c,t-1}$ -0.0564*** -0.646*** -0.0422*** -0.743*** 0.08000.003490.00425-0.0394(0.00496)(0.110)(0.00653)(0.187)(0.0148)(0.112)(0.0132)(0.170)BankCrisis_{ct} -3.161*** -2.192*** -2.797*** 5.115** -3.182*** -2.347*** -2.221*** 5.429** (0.435)(0.690)(2.462)(0.556)(2.419)(0.539)(0.453)(0.750)0.00911*** $MonPol_{c,t}$ 0.303*** 0.0119*** 0.266**0.1000.001030.0002210.102(0.00234)(0.112)(0.00240)(0.259)(0.00492)(0.109)(0.00445)(0.218) $MonPol_{c,t-1}$ 0.0191*** -0.714*** 0.0146*** -0.569** -0.00142***-0.325** -0.000605** -0.435*** (0.00180)(0.124)(0.00231)(0.238)(0.000295)(0.0973)(0.000437)(0.157) $MonPol_{c,t-2}$ 0.00108*** 0.348***0.00144***0.231**-0.225*** -0.0003802.39e-06-0.0642(0.000375)(0.0979)(0.000445)(0.114)(0.000281)(0.0824)(0.000258)(0.104)Ν 28,012 3,740 15,765 8,507 26,906 15,238 8,135 3,533 R^2 (overall) 0.2070.2710.2210.1430.2050.2790.2200.131

Robust standard errors in (). Clusters by industry-country.

^{***, **, *} denote 1%, 5%, 10% statistical significance.

Table B.3: Effects of banking crises on industrial growth, with controls for monetary, fiscal and liquidity policies

Controls Without accounting for monetary policy With monetary policy increases.

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$inflation_{c,t} = \begin{pmatrix} 0.646 & (0.873) & (1.485) & (2.055) & (1.121) & (2.073) & (1.929) & (5.632) \\ -0.00352^{***} & -0.131^* & -0.00325^{***} & -0.145^{***} & -0.0177 & 0.543^{***} & -0.0115 & 0.128 \\ (0.000412) & (0.0697) & (0.000424) & (0.0401) & (0.0191) & (0.120) & (0.0177) & (0.212) \\ inflation_{c,t-1} & 0.000242 & -0.00999 & 0.000975^{**} & 0.0335 & 0.0138 & -0.769^{***} & 0.0149 & 0.0193 \\ & & & & & & & & & & & & & & & & & & $
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$inflation_{c,t-1} \\ BankCrisis_{c,t} \\ (0.422) \\ MonPol_{c,t} \\ \\ (0.000412) \\ (0.0697) \\ (0.000424) \\ (0.000424) \\ (0.000424) \\ (0.000424) \\ (0.000478) \\ (0.0119) \\ (0.000483) \\ (0.000483) \\ (0.0272) \\ (0.0272) \\ (0.0155) \\ (0.0155) \\ (0.0155) \\ (0.0155) \\ (0.114) \\ (0.0143) \\ (0.0143) \\ (0.186) \\ (0.422) \\ (0.521) \\ (0.788) \\ (0.772) \\ (0.788) \\ (0.498) \\ (0.498) \\ (0.401) \\ (0.00450) \\ (0.00515) \\ (0.101) \\ (0.00481) \\ (0.226) \\ \\ (0.226)$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$BankCrisis_{c,t} = \begin{pmatrix} (0.000478) & (0.0119) & (0.000483) & (0.0272) & (0.0155) & (0.114) & (0.0143) & (0.186) \\ -2.488^{***} & -2.421^{***} & -1.643^{***} & -0.0909 & -3.197^{***} & -2.420^{***} & -1.694^{**} & 5.234^{**} \\ (0.422) & (0.521) & (0.788) & (1.498) & (0.464) & (0.558) & (0.772) & (2.809) \\ MonPol_{c,t} & & & & & & & & & & & & & & & & & & &$
$BankCrisis_{c,t} \begin{array}{ccccccccccccccccccccccccccccccccccc$
$MonPol_{c,t} \qquad (0.422) \qquad (0.521) \qquad (0.788) \qquad (1.498) \qquad (0.464) \qquad (0.558) \qquad (0.772) \qquad (2.809) \\ 0.00450 0.294^{***} 0.00382 0.136 \\ (0.00515) (0.101) (0.00481) (0.226)$
$MonPol_{c,t}$ $0.00450 0.294^{***} 0.00382 0.136$ $(0.00515) (0.101) (0.00481) (0.226)$
(0.00515) (0.101) (0.00481) (0.226)
M = D + 1
$MonPol_{c,t-1}$ $-0.000617^{**} -0.452^{***} -0.00141^{***} -0.359^{**}$
(0.000303) (0.0944) (0.000429) (0.172)
$MonPol_{c,t-2}$ -0.000560^* -0.214^{***} -0.000196 -0.0530
(0.000296) (0.0802) (0.000280) (0.125)
N 38,257 18,898 12,161 7,198 24,336 14,029 7,313 2,994
R^2 (overall) 0.153 0.265 0.150 0.093 0.224 0.308 0.231 0.146

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

Table B.4: Effects of banking crises on industrial growth, with interactions for the Great Financial Crisis (2007-2009)

Controls	Without accounting for monetary policy				With monetary policy increase:			
					1	$MonPol_{c,t} =$	$=\Delta MPR_{c,t}$	
	All	Advanced	Emerging	Low	All	Advanced	Emerging	Low
	countries	economies	\max	income	countries	economies	\max	income
$EFD_i \times Bank$	-2.575**	-0.900	-7.745***	-4.201	-2.086	0.673	-9.399***	-9.262**
$Crisis_{c,t}$	(1.244)	(1.682)	(1.911)	(3.236)	(1.482)	(1.744)	(2.373)	(3.802)
$EFD_i \times Bank$	-2.856*	0.118	-7.455*	-43.82	-0.850	-0.665	-5.618	
$Crisis_{c,t} \times GFC_t$	(1.469)	(1.668)	(4.032)	(43.88)	(1.523)	(1.697)	(4.169)	
$ShareManVA_{i,c,t-1}$	-0.164***	-0.152***	-0.164***	-0.173**	-0.286***	-0.233***	-0.223*	-0.423***
	(0.0359)	(0.0439)	(0.0604)	(0.0780)	(0.0572)	(0.0618)	(0.126)	(0.163)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.723***	-7.666***	-8.765***	-1.337	-10.27***	-14.15***	-13.01***	-14.54***
,	(0.596)	(0.820)	(1.300)	(1.966)	(0.983)	(1.857)	(1.616)	(4.698)
$inflation_{c,t}$	-0.00357***	-0.148**	-0.00341***	-0.148***	-0.00449	0.513***	0.00274	0.114
	(0.000421)	(0.0687)	(0.000434)	(0.0394)	(0.0182)	(0.122)	(0.0161)	(0.201)
$inflation_{c,t-1}$	0.000291	-0.00541	0.000999**	0.0300	0.00347	-0.743***	0.00394	-0.0394
	(0.000471)	(0.0116)	(0.000477)	(0.0282)	(0.0148)	(0.112)	(0.0133)	(0.170)
$BankCrisis_{c,t}$	-2.379***	-2.259***	-1.620**	0.718	-3.198***	-2.380***	-2.160***	5.429**
	(0.401)	(0.527)	(0.719)	(1.350)	(0.457)	(0.568)	(0.741)	(2.419)
$MonPol_{c,t}$					0.00103	0.266**	0.000158	0.102
					(0.00492)	(0.109)	(0.00446)	(0.218)
$MonPol_{c,t-1}$					-0.000595**	-0.431***	-0.00137***	-0.325**
					(0.000297)	(0.0979)	(0.000442)	(0.157)
$MonPol_{c,t-2}$					-0.000380	-0.222***	1.04e-05	-0.0642
					(0.000281)	(0.0824)	(0.000260)	(0.104)
N	41,545	20,228	13,286	8,031	26,906	15,238	8,135	3,533
R^2 (overall)	0.144	0.244	0.143	0.090	0.205	0.279	0.220	0.131

Robust standard errors in (). Clusters by industry-country.

***, **, * denote 1%, 5%, 10% statistical significance.

Table B.5: Effects of different types of financial crises (banking, currency, sovereign debt, any financial crisis), across the entire country sample

Controls

With macroeconomics controls

$\operatorname{Controls}$		With macro	economics co	ntrols
$\overline{EFD_i \times Banking}$				-2.893***
$Crisis_{c,t}$				(1.097)
$EFD_i \times Currency$	-5.058***			-3.913**
$Crisis_{c,t}$	(1.760)			(1.824)
$EFD_i \times Sovereign$		-2.826**		-1.719
$DebtCrisis_{c,t}$		(1.375)		(1.341)
$EFD_i \times Any$			-2.893***	
$Crisis_{c,t}$			(0.905)	
$ShareManVA_{i,c,t-1}$	-0.163***	-0.164***	-0.168***	-0.165***
	(0.0357)	(0.0358)	(0.0361)	(0.0360)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-6.043***	-6.628***	-6.891***	-6.384***
,	(0.584)	(0.595)	(0.601)	(0.600)
$inflation_{c,t}$	-0.00240***	-0.00351***	-0.00336***	-0.00234***
	(0.000409)	(0.000427)	(0.000418)	(0.000415)
$inflation_{c,t-1}$	-0.000125	0.000485	0.000647	5.70e-05
	(0.000488)	(0.000463)	(0.000457)	(0.000475)
$BankCrisis_{c,t}$				-1.777***
				(0.408)
$CurrencyCrisis_{c,t}$	-4.857***			-4.431***
	(0.795)			(0.822)
$DebtCrisis_{c,t}$		-0.997**		-0.365
		(0.481)		(0.489)
$AnyCrisis_{c,t}$			-2.279***	
			(0.365)	
N	41,545	41,545	41,545	41,545
R^2 (overall)	0.145	0.140	0.144	0.148
D 1	1 1	() (1)		

Robust standard errors in (). Clusters by industry-country. ***,**,* denote 1%, 5%, 10% statistical significance.

Table B.6: Effects of different types of financial crises across country groups Controls With macro controls

Controls		***************************************	madio comercia
	Advanced	Emerging	Low
	economies	\max	income
$\overline{EFD_i \times Banking}$	-1.072	-7.862***	-2.196
$Crisis_{c,t}$	(1.298)	(2.035)	(3.415)
$EFD_i \times Currency$	4.844	-2.605*	-9.921
$Crisis_{c,t}$	(3.686)	(1.542)	(6.123)
$EFD_i \times Sovereign$	-4.158	-0.0822	-1.289
$DebtCrisis_{c,t}$	(5.177)	(1.624)	(2.164)
$ShareManVA_{i,c,t-1}$	-0.153***	-0.162***	-0.175**
	(0.0439)	(0.0598)	(0.0795)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-7.761***	-9.132***	-1.422
,	(0.826)	(1.340)	(1.946)
$inflation_{c,t}$	-0.144**	-0.00232***	-0.115***
	(0.0689)	(0.000433)	(0.0414)
$inflation_{c,t-1}$	-0.00217	0.000837*	0.0116
	(0.0114)	(0.000468)	(0.0282)
$BankCrisis_{c,t}$	-1.983***	-0.695	0.0482
	(0.506)	(0.763)	(1.425)
$CurrencyCrisis_{c,t}$	-5.452***	-4.306***	-2.043
	(1.761)	(0.779)	(2.496)
$DebtCrisis_{c,t}$	-1.643	-2.284***	1.208
	(1.455)	(0.695)	(0.881)
N	20,228	13,286	8,031
R^2 (overall)	0.245	0.149	0.092
Dalinat atan	11	· · · () Cl · · · · ·	ong has in deadans consistent

Robust standard errors in (). Clusters by industry-country.

***,**,* denote 1%, 5%, 10% statistical significance.

Table B.7: Quantile regressions with country-industry fixed effects for the impact of different types of financial crises (banking, currency, sovereign debt)

, ·	1	All countri	es	Advanced	Emerging	Ĺow
				economies	\max	income
Controls	Q25	Q50	Q75	Q50	Q50	Q50
$EFD_i \times Banking$	-3.904	-3.003	-2.163	-1.359	-7.591	-2.297
$Crisis_{c,t}$	(3.544)	(2.076)	(1.537)	(5.070)	(14.20)	(4.548)
$EFD_i \times Currency$	-3.722	-3.745	-3.767	4.664	-2.669	-10.18*
$Crisis_{c,t}$	(5.671)	(3.323)	(2.460)	(16.17)	(11.75)	(5.445)
$EFD_i \times Sovereign$	-3.589	-1.767	-0.0667	-4.054	-0.00222	-1.393
$DebtCrisis_{c,t}$	(4.363)	(2.557)	(1.893)	(19.11)	(10.93)	(2.730)
$ShareManVA_{i,c,t-1}$	-0.0680	-0.173***	-0.270***	-0.171	-0.173	-0.180**
	(0.108)	(0.0635)	(0.0470)	(0.132)	(0.380)	(0.0827)
$\ln(GDP_{c,t-1}^{PPP,pc})$	-4.111***	-6.856***	-9.417***	-7.869***	-8.090	-4.188**
3,0 1	(1.535)	(0.900)	(0.667)	(1.976)	(6.700)	(1.841)
$inflation_{c.t}$	-0.00378	-0.00195	-0.000253	0.0511	-0.00173	-0.0988**
,	(0.00362)	(0.00212)	(0.00157)	(0.187)	(0.00644)	(0.0493)
$inflation_{c,t-1}$	-0.000568	-0.000469	-0.000376	-0.0503	-0.000304	-0.0131
,	(0.00197)	(0.00115)	(0.000854)	(0.0839)	(0.00348)	(0.0420)
$BankCrisis_{c,t}$	-4.070***	-2.541***	-1.115*	-3.197	-1.309	-0.0661
,	(1.371)	(0.804)	(0.595)	(1.975)	(5.222)	(1.675)
$CurrencyCrisis_{c,t}$	-5.865***	-4.417***	-3.066***	-3.687	-4.852	-2.357
,	(2.121)	(1.243)	(0.921)	(7.246)	(4.353)	(1.984)
$DebtCrisis_{c,t}$	0.0647	-0.533	-1.090	-2.920	-2.256	1.849*
	(1.646)	(0.965)	(0.714)	(5.736)	(4.363)	(1.084)
$GreatModeration_t$	-1.216*	-1.051***	-0.896***	-1.471*	-0.857	-1.125
(1988-2006)	(0.693)	(0.406)	(0.301)	(0.830)	(3.153)	(1.106)
GFC_t	-7.031***	-4.439***	-2.020***	-5.936***	-3.864	-1.912*
(2007-2009)	(0.982)	(0.576)	(0.428)	(1.467)	(3.305)	(1.110)
N	41,545	41,545	41,545	20,228	13,286	8,031

Standard errors in ().

***, **, * denote 1%, 5%, 10% statistical significance.

Table B.8: Effects of different types of financial crises with industry-country, country-year and industry-year fixed effects Controls All Advanced Emerging Low

			0 0	
	${\rm countries}$	economies	\max	income
All crises	in the sar	ne regressi	on model	
$EFD_i \times Banking$	-2.612***	-1.479	-8.134***	-1.605
$Crisis_{c,t}$	(1.011)	(1.618)	(1.766)	(3.603)
$EFD_i \times Currency$	-4.133**	1.381	-2.470	-8.980
$Crisis_{c,t}$	(1.964)	(4.340)	(1.798)	(6.419)
$EFD_i \times Sovereign$	-0.972	-0.693	-1.878	0.107
$DebtCrisis_{c,t}$	(1.107)	(3.720)	(1.694)	(2.126)
$ShareManVA_{i,c,t-1}$	-0.158***	-0.158***	-0.142***	-0.182**
, ,	(0.0341)	(0.0503)	(0.0544)	(0.0784)
N	41,910	20,400	13,470	7,981
R^2 (overall)	0.343	0.426	0.414	0.292
Coefficients obtained	l with diffe	erent types	of crises in	n separate

Coefficients obtained with different types of crises in separate regressions

$EFD_i \times Banking$	-3.183***	-1.425	-8.758***	-2.671
$Crisis_{c,t}$	(0.986)	(1.579)	(1.814)	(3.574)
$EFD_i \times Currency$	-4.944***	0.597	-4.130**	-9.171
$Crisis_{c,t}$	(1.894)	(4.244)	(1.850)	(6.268)
$EFD_i \times Sovereign$	-1.907*	-1.012	-2.976*	-0.555
$DebtCrisis_{c,t}$	(1.094)	(3.679)	(1.718)	(2.110)
$EFD_i \times Any$	-2.533***	-1.592	-5.503***	-1.569
$Crisis_{c,t}$	(0.867)	(1.439)	(1.537)	(3.108)
N	41,910	20,400	13,470	7,981

All regressions include $ShareManVA_{i,c,t-1}$ as a control, but its coefficient is omitted in the regressions with each type of crisis separately.

Clusters by industry-country and industry-year. Robust standard errors in ().

***, **, * denote 1%, 5%, 10% statistical significance. Fixed effects in the regressions are omitted.

Table B.9: Effect on total manufacturing growth (in %) of different types of

	financial crises (average across countries in 2019): estimates from all the models											
All crises in the same regression model												
	Coefficient average	Tripe FE (i,c+i,t+c,t) $\gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$			Double FE $(i,c+c,t)$			FE (i,c+t) plus controls				
	across countries				$\gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$			$\delta_{Crisis} + \gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$				
	Crisis type	Bank	Currency	Debt	Bank	Currency	Debt	Bank	Currency	Debt		
	All countries	-0.8	-1.3	-0.3	-1.0	-1.3	-0.3	-2.7	-5.7	-0.9		
	AEs	-0.5	0.5	-0.2	-0.4	0.4	-1.3	-2.3	-3.9	-3.0		
	EMs	-2.8	-0.9	-0.7	-2.6	-0.8	0.0	-3.4	-5.2	-2.3		
	LICs	-0.5	-2.6	0.0	-0.9	-2.6	-0.3	-0.6	-4.9	0.8		

Separate models for each crises' type

Coefficient average	Tripe FE (i,c i,t c,t)			Double FE $(i,c+c,t)$			FE (i,c+t) plus controls		
across countries	$\gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$			$\gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$			$\delta_{Crisis} + \gamma \sum_{i} \varpi_{i,c,t} EFD_{i}$		
Crisis type	Bank	Currency	Debt	Bank	Currency	Debt	Bank	Currency	Debt
All countries	-1.0	-1.6	-0.6	-1.1	-1.6	-0.7	-3.4	-6.5	-1.9
AEs	-0.5	0.2	-0.3	-0.4			-2.5		
EMs	-3.1	-1.4	-1.0	-2.8			-4.6		
LICs	-0.8	-2.6	-0.2	-1.6			-1.1		

Table B.9 focuses on the estimates obtained from the triple fixed effect regressions (Table B.8), the double fixed effect regressions (Tables B.1, B.5 and B.6) and the fixed effects model with macroeconomic controls (Tables 3, 7 and 8). For simplicity, I show only the effects for the last year of the sample, which corresponds to the last pre-pandemic year, 2019. The results for the triple and the double fixed effect models are almost the same, both qualitatively and in size. Therefore, I will comment only the triple fixed effect results and then the total effect from the model with macroeconomic controls. The results are also similar, whether qualitative or in size, if one considers the simultaneous regression with all crises or the separate regressions. Therefore, I will comment only the results from the simultaneous regressions.

The estimates from the coefficients on total manufacturing due to the external finance dependence of industries $(\gamma \sum_i \varpi_{i,c,t} EFD_i)$ show that EMs are much more affected by banking crises, while LICs are more affected by currency crises. From the models with all crises in the same regression,

the triple FE estimates show that the average country experiences a manufacturing growth loss of 0.8%, 1.3% and 0.3% during banking, currency and sovereign debt crises. For AEs, there is a reduction in manufacturing growth of 0.5% and 0.2% during banking and sovereign debt crises. For LICs there is a reduction in growth of 0.5% and 2.6% during banking and currency crises.

Finally, considering the total effect from the model with fixed effects and macroeconomic controls, it is shown that currency crises have the strongest impact. There is a loss in industrial growth of 2.7%, 5.7% and 0.9% during banking, currency and sovereign debt crises, in the all countries sample. Across country groups, it is shown that EMs experience the strongest effects from both banking and currency crises, while AEs experience the strongest effect of sovereign debt crises. Perhaps the stronger effect of sovereign debt crises in AEs is due to their larger sovereign debt markets and due to these crises being more unexpected in such countries. EMs and LICs may be less affected by sovereign debt crises, because their use of sovereign debt is smaller. In AEs there is a reduction in industrial growth of 2.3%, 3.9% and 3% during banking, currency and debt crises. In EMs there is a loss of industrial growth of 3.4%, 5.2% and 2.3% during banking, currency and sovereign debt crises. Finally, LICs experience an industrial growth loss of 0.6% and 4.9% during banking and currency crises. LICs are the countries least affected by banking crises, which could be due to the low development of their banking sector (Rajan and Zingales 1998, Raddatz 2006).

Documentos de Trabajo Banco Central de Chile

NÚMEROS ANTERIORES

La serie de Documentos de Trabajo en versión PDF puede obtenerse gratis en la dirección electrónica:

www.bcentral.cl/esp/estpub/estudios/dtbc.

Existe la posibilidad de solicitar una copia impresa con un costo de Ch\$500 si es dentro de Chile y US\$12 si es fuera de Chile. Las solicitudes se pueden hacer por fax: +56 2 26702231 o a través del correo electrónico: bcch@bcentral.cl.

Working Papers Central Bank of Chile

PAST ISSUES

Working Papers in PDF format can be downloaded free of charge from:

www.bcentral.cl/eng/stdpub/studies/workingpaper.

Printed versions can be ordered individually for US\$12 per copy (for order inside Chile the charge is Ch\$500.) Orders can be placed by fax: +56 2 26702231 or by email: bcch@bcentral.cl.

DTBC - 1044

The impact of financial crises on industrial growth: lessons from the last 40 years

Carlos Madeira

DTBC - 1043

Heterogeneous UIPDs Across Firms: Spillovers from U.S. Monetary Policy Shocks

Miguel Acosta-Henao, Maria Alejandra Amado, Montserrat Marti, David Perez-Reyna

DTBC - 1042

Bank Competition and Investment Costs across Space

Olivia Bordeu, Gustavo González, Marcos Sorá

DTBC - 982*

Freight Costs and Substitution Among Import Regions: Implications for Domestic Prices

Gustavo González, Emiliano Luttini, Marco Rojas

DTBC - 1041

Tail-Risk Indicators with Time-Variant Volatility Models: the case of the Chilean Peso Rodrigo Alfaro, Catalina Estefó DTBC - 1040

Exposición de la Banca en Chile a los Riesgos Financieros relacionados al Clima: Riesgos Físicos y de Transición

Luis Gonzales, Cristian Rojas

DTBC - 971*

Production Network Formation, Trade, and Welfare

Costas Arkolakis, Federico Huneeus, Yuhei Miyauchi

DTBC - 1011*

Macro Implications of Inequality-driven Political Polarization

Alvaro Aguirre

DTBC - 967*

Firm Financing During Sudden Stops: Can

Governments Substitute Markets?

Miguel Acosta-Henao, Andrés Fernández, Patricia Gomez-Gonzalez, Sebnem Kalemli-Ozcan

DTBC - 1039

The Effects of Educated Leaders on Policy and Politics:

Quasi-Experimental Evidence from Brazil

Paulo Bastos, Cristián Sánchez

DTBC - 1038

Equilibrium Consequences of Vouchers Under Simultaneous Extensive and Intensive Margins Competition

Cristián Sánchez

DTBC - 1037

Dinámicas del Mercado Laboral Formal en Chile:

Evidencia desde los Microdatos

Roberto Gillmore, Gustavo González, Nicolás Rivera

DTBC - 1036

Micro-Evidence on the Consumption Impact of Income-Support Policies During COVID-19

Metodij Hadzi-Vaskov, Emiliano Luttini, Luca Antonio Ricci

